The Rights of Persons with Disabilities Bill, 2011

Presented by Committee appointed by

Ministry of Social Justice and Empowerment

Government of India

30 June 2011

Prepared by

Centre for Disability Studies NALSAR University of Law Hyderabad, India 30 June 2011

Contents

Sec.	Title	Page
PART	1: REPORT & ANNEXURES	
1	Final Report	3
	Annexures	
	List of Committee Members	11
	Committee Meetings Held	18
	National Consultations with Civil Society Representatives	39
	State level consultations in 28 states and 2 UTs	42
	Legal Consultation	49
	Subgroup Meeting reports	60
	Acknowledgments	75
PART	2: THE BILL	
2	Justificatory Note on Proposed Bill	78
3	The Rights of Persons with Disabilities Bill, 2011	83
	Part I: Rights and Entitlements	98
	Part 2: Powers, Duties and Responsibilities	145
	Part 3: Regulatory and Monitoring Authorities	156
	Part 4: Grievance Redressal	181
	Part 5: Offences and Penalties	188
	Part 6: Miscellaneous	193
	Schedule 1: List of Disabilities	201

Part 2

The Rights of Persons with Disabilities Bill, 2011

JUSTIFICATORY NOTE ON PROPOSED BILL

Construction of the Terms of Reference

The proposed bill recognizes the equality of persons with disabilities and prohibits direct or indirect discrimination on the basis of disability. The Bill also makes provision for the practical implementation of all the civil political rights included in the Convention on the Rights of Persons with Disabilities (CRPD). Many of these rights are also affirmed by the Constitution of India. The inclusion of provisions on these rights in the Bill is in accordance with the terms of reference given to the Committee as the Bill, unlike the Constitution, is not the source of these rights for persons with disabilities. The Bill is the implementation primer of these rights as it details out how these rights should be realized for persons with disabilities. The Right to Information Act and the Right to Education Act of 2009 were enacted with similar objectives.

The Constitution of India includes persons with disabilities generically within the universal categories of person and citizen; it does not even mention disability in the prohibited grounds specified in articles 15 and 16. Consequently, when the Bill lays down procedures by which the civil-political rights guaranteed in the CRPD and the Constitution of India will be realized for persons with disabilities, it provides for matters which are not addressed in the Constitution. The rights guaranteed by the Constitution need adaptation to be of value for persons with disabilities. For example, the right to equality and non-discrimination is of little value to persons with disabilities without accessibility and reasonable accommodation. The freedom of speech and expression holds little meaning without recognition of the means, modes and formats of communication practiced by persons with disabilities. The Bill undertakes this exercise in order to implement the CRPD and the Constitutional mandate for persons with disabilities. Consequently, the provisions of the Bill are distinct from and not a duplication of the provisions of the Constitution.

Preference to Generic over Specific Language

Article 4 (3) of the CRPD requires that there shall be close consultation with persons with disabilities through their representative organizations for making any legislation to implement the

convention. The Committee's report outlines in detail this process of consultation. The text of the legislation explicitly footnotes the suggestions received at the various state consultations as also in the legal consultation and the manner in which the working text of the proposed bill was altered to incorporate the various suggestions received. Insofar as the suggestions put forth at the state consultations were informed by the lived experience of persons with disabilities, they were often, though not always, very specific in their import. Keeping in view the inflexible nature of the legal text, these specific suggestions have been converted into generic ones. Thus for example, the naming of every kind of transportation, public building or service has been avoided and generic terminology used. In some parts of the legislation however illustrative naming has been practiced. In the main, this illustrative naming has been used for those groups, services or institutions or areas who may be lost within generic language unless explicitly named. Thus, for example, jails, prisons and places of detention have been explicitly named in the definition of public building. On a similar reasoning, the requirements of persons with disabilities in rural areas, tribal regions, hilly areas and conflict zones have been explicitly named.

Principles Guiding Inclusion in Legislation, Rules, Schemes and Programs

The mandatory nature of law makes it an attractive medium for ensuring entitlements. Consequently, both during the civil society consultations prior to the working draft; as also in the state consultations on the working draft; a number of memorandums were received which sought very particular protections in the text of the new legislations; or made very specific suggestions on the procedure of implementation. In order to respect the sentiment behind these suggestions, without making their acceptance a stranglehold on future development, this legislation has been guided by the following principles of incorporation. The rights and the principles informing the implementation of the rights have been incorporated in the text of the law. Matters of detailed procedure which may need revision have been left to be addressed in the rules. The guaranteeing of entitlements is in the text of the law; the details of the scheme or programme through which the entitlement shall be implemented has been left to be notified in the particular scheme or program. This has been done to avoid the dangers of both over prescription and excessive flexibility and to obtain the optimum degree of compulsion and flexibility.

Nature of Duty Bearer

The Supreme Court of India has in its interpretations of Article 12 of the Indian Constitution continually underlined the fact that the entity of State is much wider than just the central, state and local governments. Hence, the duty to respect the rights guaranteed under Part III of the Indian Constitution does not only reside in these governments alone. In relation to practices of social discrimination, the Constitution of India has even placed obligations on the people. Thus, the Constitution has horizontal application when it addresses the issue of social exclusion. The CRPD also requires state parties to "take all appropriate measures to eliminate discrimination on the basis of disability by any person, organization or private enterprise". Keeping in view, the mandate of the CRPD and the text and jurisprudence of the Indian Constitution, the proposed legislation makes appropriate governments, concerned establishments and individual persons, as the case may be, responsible for fulfilling the obligations of the statute.

Judicial Precedents and Juristic Theory as Source of Law

Indian and foreign judicial precedents; non-discrimination legislation of other countries and juristic theory has been drawn upon to shape the content of the proposed Bill. Thus, for example, the equality and non-discrimination provision especially the definition of reasonable accommodation has been constructed after studying the manner in which the concept has been interpreted by courts of countries which have incorporated the concept in their domestic legislation. The legislative provisions and judicial decisions which deny certain rights such as legal capacity to persons with disabilities antedate disability rights jurisprudence. These precedents were made in the absence of persons with disabilities and hence are to that extent decisions arrived at without providing a hearing to an essential party in the matter. In the face of this reality certain parts of the proposed legislation have only drawn from the text of the CRPD; the juristic writings on the CRPD and the pronouncements of the Committee on the Rights of Persons with Disabilities which is the authoritative body of the United Nations on the nature of the obligation under the Convention.

The New Disability Rights Paradigm

The CRPD has ushered in a new paradigm of disability rights. The social model of disability; universal legal capacity with support and no forced interventions are critical components of this new paradigm. The proposed Bill recognizes all three features of the new paradigm in its text and makes a commitment to make them the driving force of disability rights. Consequently it mandates a range of ground level initiatives such as the development of a socio-medical scale; regimes of support and alternatives to institutions so that the new paradigm in fact informs the lived reality of persons with disabilities. As these alternatives do not exist at present, to guard against a situation of a legal vacuum, it makes a range of transitional arrangements on questions such as definition of disability; guardianship and compulsory care. These transitional arrangements have been made in such manner as to ensure that the values of the new paradigm even inform the transitional period.

Status of Jammu and Kashmir

Article 370 of the Constitution accords a distinct status to the state of Jammu & Kashmir in matters of law making. It is due to this constitutional status that the standard statutory formulation is that a statute "extends to the whole of India except the state of Jammu and Kashmir". The consultations on the working draft of this statute were also undertaken in the state of Jammu and Kashmir. In view of this active participation, Clause 1 (2) has departed from the standard statutory formula of referring to the State of J&K and drafted this clause by specifying the constitutional procedure by which this law could be extended to Jammu and Kashmir.

CRPD as Guiding Principle

The statement of objects and reasons outline that the legislation has been enacted in order to bring the Indian Disability Law in harmony with the CRPD. However, the statement of objects and reasons is examined by Courts only when there is an ambiguity in the text of the law. Consequently in order to ensure that the CRPD shall operate as a guiding principle for both administrators and adjudicators clause 3 has been included. This clause has been explicitly

incorporated to underscore that the CRPD should inform implementation and adjudication at all times and should not be only looked at when there is an ambiguity in the text of the law.

Structure of the Law

Other than preliminary part, the legislation has been divided into six parts. Part I contains the rights and entitlements; Part II provides for powers, duties and responsibilities; Part III elaborates on regulatory and monitoring authorities; Part IV addresses grievance redressal; Part V deals with offences and penalties; and Part VI is miscellaneous segment which is followed by a schedule on the named list of disabilities which accompanies the generic definition.

Dr. Amita Dhanda

Professor and Head

Centre for Disability Studies

18 henda.

NALSAR, Hyderabad

THE RIGHTS OF PERSONS WITH DISABILITIES BILL, 2011

		Page
State	ment of Objects and Reasons	89
Prean	nble	90
PREL	IMINARY	
1	Short Title Extent and Commencement	91
2	Definitions	91
3	Guiding Principle for Implementation and Interpretation	98
PART	I: RIGHTS AND ENTITLEMENTS	
4	Right to Equality and Non discrimination	98
5	Pro-active Interventions for Persons with Disabilities with increased vulnerability.	99
6	Women with Disabilities	100
7	Right to Education of Women and Girls with Disabilities	101
8	Right to Work and Employment of Women with Disabilities	101
9	Right of Women and Girls with Disabilities to be protected against	102
all for	ms of Abuse, Violence and Exploitation	
10	Right of Women and Girls with Disabilities to Health	103
11	Right of Women with Disabilities to Home and Family	104
12	Access to Justice for Women with Disabilities	104
13	Obligations of the National Commission for Women	104
14	Rights of Children with Disabilities	105
15	Rights of Children with Disabilities to Health	106
16	Rights of Children with Disabilities to Leisure, Culture and Sports	106
17	Access to Justice for Children with Disabilities	106
18	Right to Legal Capacity and Equal Recognition before the Law	106
19	Replacement of Plenary Guardianship with limited Guardianship	108
20	Duty to Provide Support	108
21	Denial of Legal Capacity	109
22	Right to Life and Living	109
23	Right to Liberty	110
24	Right to Live in the Community	111
25	Right to Integrity	112
26	Protection from Abuse, Violence and Exploitation	113
27	Protection and Safety of Persons with Disabilities in Situations of Risk	115
28	Right to Home and Family	116

		Page
29	Additional Provision for Disability Related Needs in Maintenance and Alimony	117
30	Reproductive of Rights of Persons with Disabilities	117
31	Freedom of Speech and Expression	118
32	Right to Political Participation	118
33	Access to Justice	119
34	Definitions for Right to Education	121
35	Right to Education	123
36	Surveying for Children with Disabilities	123
37	Establishment of Neighbourhood Schools, Resource Centres and Special Schools	123
38	Right to Free Childhood Care and Pre-school Education	124
39	Entitlements of Children with Disabilities	125
40	Reasonable Accommodation in Education	125
41	Student Teacher Ratio for Children with Disabilities	126
42	Right to School Admission	126
43	Right to Support and Accessibility	126
44	No Detention	127
45	Impermissible Disciplinary Measures	127
46	Parents of Children with Disabilities in School Management Committees	127
47	Disability Related Needs in School Development Plans	127
48	Education Reform Commission for Advancement of Disability Rights Perspective	128
49	Right to Higher Education	128
50	Reservation in Higher Educational Institutions	129
51	Support in Higher Education	129
52	Qualifications of Educators	129
53	No Capitation Fee in Higher Education	130
54	Leisure, Culture Sport in Schools and Higher Educational Institutions	130
55	Adult Education for Persons with Disabilities	130
56	Non-discrimination in Employment	130
57	Reservations	132
58	Equal Opportunity Policies	133
59	Maintenance of Records	133
60	Vocational Training, Rehabilitation and Self Employment	134
61	Incentives to Establishments	135

		Page
62	Grievance Redressal	135
63	Special Employment Exchanges	135
64	Right to Social Security	136
65	Right to Health	138
66	Insurance	139
67	Ethical Guidelines	140
68	Awareness Raising Responsibilities	140
69	Prohibition on Denial of Food and Fluids	141
70	Right to Habilitation	141
71	Right to Rehabilitation	142
72	Research and Development	143
73	Right to Leisure, Culture and Recreation	143
74	Right to Participation in Sports, Games and Athletics	144
PART	II: POWERS, DUTIES AND RESPONSIBILITIES	
75	Awareness Raising	145
76	Accessibility	147
77	Accessibility Standards	147
78	Personal Mobility	148
79	Access to Services	149
80	Access to Information and Communication Technology	149
81	Access to Consumer Goods and Services	150
82	Service Animals	150
83	Mandatory Observance of Accessibility Norms	151
84	Time limit for making existing Infrastructure and Premises Accessible	151
85	Time limit for Accessibility and Duties of Service Provider	151
86	Role of the National Centre for Universal Design and Barrier-Free	152
	Environment	
87	Implementation Plan	152
88	Human Resource Development	153
89	Financial Assistance to Registered Organizations	154
90	Disability Cell	155
91	Disability Audit	155
92	Power to Amend Schedule	155
93	Disability Certificates	155

		Page
PART	III: REGULATORY AND MONITORING AUTHORITIES	
94	Establishment and Incorporation of National Disability Rights Authority	156
95	Composition and Management of the Authority	156
96	Powers, Functions and Meetings of the Governing Board	158
97	Executive Board	159
98	Powers, Functions and Meetings of the Executive Board	159
99	Reservation within the Elected Members of the Governing Board	160
100	Selection Committee for the Chairperson	161
101	Term of Officer of Chairperson and Members	161
102	Terms and Conditions of Chairperson and Members	162
103	Officers and other Staff of National Disability Rights Authority	162
104	Resignation, Removal and Suspension of the Chairperson and other Members.	162
105	Functions of the National Disability Rights Authority	163
106	Power to Conduct Enquiry	165
107	Power to Issue Directions	166
108	Power to Review and Advice	167
109	Power to Review, Advice and Report	168
110	Power to Associate Individuals and Organizations	168
111	Power to Grant Exemptions	168
112	Establishment of Committees	169
113	Power of National Disability Rights Authority vis-à-vis the State Disability Rights Authority	169
114	Annual Report of the National Disability Rights Authority	170
115	Establishment Incorporation of State Disability Rights Authority	170
116	Composition and Management of State Disability Rights Authority	171
117	Selection Committee for the Chairperson of State Disability Rights Authority	172
118	Term of office of Chairperson and Members of State Disability Rights Authority	173
119	Terms and Conditions of Service of Chairperson and Members of State Disability Rights Authority	173
120	Officers and other staff of State Disability Rights Authority	174
121	Resignation, Removal and Suspension of Chairperson and Other Members of State Disability Rights Authority	174
122	Application of Certain Provisions Relating to National Disability Rights	175

		Page
	Authority to State Disability Rights Authority	
123	Duties of State Disability Rights Authority vis-à-vis National Disability	175
	Rights Authority	
124	Annual and Special Reports of State Disability Rights Authority	176
125	Registration of Organizations for Persons with Disabilities	176
126	Establishment of National Fund for Persons with Disabilities	177
127	Dissolution of Erstwhile Fund for Persons with Disabilities and	178
	Transfer of their Rights, Assets and Liabilities to the National Fund for	
	Persons with Disabilities	
128	Grants by Central Government	179
129	Grants by State Government	179
130	Funding Pattern	179
131	Exemption from Tax on Wealth and Income	180
132	Accounts and Audit	180
133	Accounts and Audit of State Disability Rights Authority	181
PART	IV: GRIEVANCE REDRESSAL	
134	District Disability Rights Tribunal	181
135	Jurisdiction of the District Disability Rights Tribunal	182
136	Panel of Disability Rights Defenders	182
137	Proceedings Before the District Disability Rights Tribunal	182
138	State Disability Rights Tribunal	183
139	Jurisdiction of the State Disability Rights Tribunal	183
140	Proceedings before the State Disability Rights Tribunal	184
141	National Disability Rights Tribunal	185
142	Jurisdiction of National Disability Rights Tribunal	185
143	Functions of the Disability Rights Tribunals	186
144	Powers of Disability Rights Tribunals	186
145	Execution of the Orders of Disability Rights Tribunals	187
146	Conditions of Service of Members of Disability Rights Tribunals	187
147	Officers and other Staff of Disability Rights Tribunals	187
D 4 5 -	TV OFFENORS AND DENALTIES	
PARI	V: OFFENCES AND PENALTIES	
1.40	Complement of Officers and Invitable of One In	100
148	Cognizance of Offences and Jurisdiction of Courts	188
149	Offences by Establishments	188
150	General Penalty for Offences	189

		Page
151	Punishment for Injury to Person with Disability	189
152	Penalty for Wrongful Medical Procedures	189
153	Penalty for Forceful Termination of Pregnancy	190
154	Penalty for Hate Speech	190
155	Penalty for Denying Food and Fluids	190
156	Penalty for Contravention of Accessibility Norms for Built Infrastructure	191
157	Penalty for Contravention of Duties by Service Providers	191
158	Penalty for Fraudulently Availing Benefit meant for Persons with Disabilities	191
159	Penalty for Willfully Making False Statements	192
160	Penalty for Failure to Furnish Information	192
161	Penalty for Failure to Comply with Directions	192
PAR	VI: MISCELLANEOUS	
162	Effect of this Act on Other Laws	193
163	Action taken in Good Faith	193
164	Provision for Mandatory Review	193
165	Removal of Difficulties Clause	194
166	Power of Central Government to make Rules	194
167	Power of State Government to make Rules	196
168	Powers of the National Disability Rights Authority to make Regulations	197
169	Powers of the State Disability Rights Authority to make Regulations	199
170	Repeal and Savings	200
SCHE	EDULE 1	
List	of Disabilities	201

Statement of Objects and Reasons

India has ratified the UN Convention on the Rights of Persons with Disabilities (UN CRPD) and undertaken the obligation to ensure and promote the full realization of all human rights and fundamental freedoms for all Persons with Disabilities without discrimination of any kind on the basis of disability. In fulfillment of this international commitment, the country is obligated to enact suitable legislation in furtherance of the rights recognized in the UN CRPD.

India enacted the Persons with Disabilities (Equal Opportunities, Protection of Rights and Full Participation) Act of 1995 in fulfillment of its obligation as a signatory to the Proclamation on the Full Participation and Equality of People with Disabilities in the Asia Pacific Region. This legislation has been on the statute book for nearly 15 years and has been the basis of an empowering jurisprudence on the rights of persons with disabilities. Whilst the need to retain the empowering jurisprudence is unequivocally acknowledged; it is also recognized that the present Persons with Disabilities Act, 1995 does not incorporate many of the rights recognized in the UN CRPD.

Furthermore, the UN CRPD recognizes that disability is an evolving concept and that disability results from the interaction between persons with impairments and attitudinal and environmental barriers that hinder their full and effective participation in society on an equal basis with others. The Persons with Disabilities Act, 1995 on the other hand has provided for an impairment based exhaustive definition of disability. Consequently, people with impairments not expressly mentioned in the Act have often been denied the rights and entitlements recognized in the Act.

It is proposed to repeal the extant legislation and replace it with a comprehensive law which is in conformity with the UN CRPD and hence this Bill.

The salient features of the proposed legislation are as follows:

a. to guarantee equality and non-discrimination to all persons with disabilities;

- b. to recognize legal capacity of all persons with disabilities and make provision for support where required to exercise such legal capacity;
- c. to recognize the multiple and aggravated discrimination faced by women with disabilities and induct a gendered understanding in both the rights and the programmatic interventions;
- d. to recognize the special vulnerabilities of children with disabilities and ensure that they are treated on an equal basis with other children;
- e. to mandate proactive interventions for persons with disabilities who are elderly, confined to their homes, abandoned and segregated or living in institutions and also those who need high support;
- f. to establish National and State Disability Rights Authorities which facilitate the formulation of disability policy and law with active participation of persons with disabilities; dismantle structural discrimination existing against persons with disabilities and enforce due observance of regulations promulgated under this Act for the protection, promotion and enjoyment of all rights guaranteed in this Act;
- g. to specify civil and criminal sanctions for wrongful acts and omissions.

Preamble

The Constitution of India resolves to secure to all its citizens justice; liberty; equality; and fraternity. Citizens with Disability are an essential part of human diversity;

India has signed and ratified the United Nations Convention on the Rights of Persons with Disabilities and thereby made an international commitment to promote, protect and ensure the rights recognized in that Convention;

Persons with Disabilities have a right:

- to integrity, dignity and respect with full participation and inclusion;
- to live a life free of shame, ridicule, or any form of disempowerment and stereotyping;
- to be entitled on an equal basis with others to all civil-political and socioeconomic rights guaranteed by international and national law.

To this end the Union of India, in its sixty-third year, enacts The Rights of Persons with Disabilities Act as follows:

PRELIMINARY

1. Short Title, Extent and Commencement

- (1) This Bill may be called the "The Rights of Persons with Disabilities Act, 2011";
- (2) It extends to the whole of India. However, its application to the State of Jammu & Kashmir shall be specified by the President of India by order only after the State of Jammu & Kashmir gives its concurrence on the application of this law to that State;
- (3) It shall come into force on such date as the Central Government may by notification appoint but in no case later than one year after its assent by the President.

2. Definitions

In this Act, unless the context otherwise requires:

(1) 'Abuse' means any act or series of acts including physical force on the body of the person with disability; or insulting, ridiculing or humiliating; or any conduct of a sexual nature that violates the dignity of the victim; or depriving with economic and financial resources; or depriving or denying support which the person with disability had demanded or which could be reasonably understood to have been demanded; with the intention to cause physical, emotional, mental, physical, or sexual injury and includes regular attempts at doing such act;

(2) 'Appropriate Government' means:-

a. in relation to the Central Government or any establishment wholly or substantially financed by that Government, or a Cantonment Board constituted under the Cantonment Act, 1924, the Central Government;

- b. in relation to a State Government or any establishment wholly or substantially financed by that Government, or any local authority, other than a Cantonment Board, any government or body constituted under the sixth schedule of the Constitution of India, the State Government;
- (3) 'Augmentative and Alternative Communication (AAC)' are alternative systems or methods of communication that may supplement or complement the use of one's own speech to fulfil the daily communication needs of persons with speech, communication or language disabilities in accordance with their requirements and enables them to participate and contribute to their community and society;
- (4) 'Barrier' means any factor that impedes, prevents or obstructs the full and effective participation, of persons with disabilities in society. This will include attitudinal, communicational, cultural, economic, environmental, institutional, political, religious, social and structural obstructions;
- (5) 'Care giver' is a person who provides paid care, support and assistance to the person with disability;
- (6) 'Communication' includes languages, display of text, Braille, tactile communication, signs, large print, accessible multimedia as well as written, audio, plain-language, human-reader and augmentative and alternative modes, means and formats of communication, including accessible information and communication technology;
- (7) 'Convention' means The United Nations Convention on the Rights of Persons with Disabilities:
- (8) 'Direct Discrimination' occurs when whether intentionally or unintentionally:
 - a. a person or group of persons to whom one or more prohibited grounds apply is treated less favourably than another person or another group of persons is, has been, or would have been treated in a comparable situation; or
 - b. for a reason which is at least in part related to one or more prohibited grounds, a person or group of persons is subjected to a disadvantage; or
 - c. there is failure to provide reasonable accommodation;
 - d. a person or a group of persons is harassed or victimized.
- (9) 'Disadvantage' means any restriction on or denial of a right, freedom,

entitlement, remuneration, goods, benefit, accessibility, service provision, opportunity, license or any other social, political, economic, cultural, spiritual, religious or civil interest; and includes any segregation of a person or a group of persons with disabilities to whom one or more prohibited grounds apply in comparison to others;

Explanation

It is immaterial that such restriction, denial or segregation is unintentional, or that it is intended for the purpose of protecting a person to who one or more prohibited grounds apply.

- (10) 'Disabled Persons' Organizations' (DPOs) means organizations controlled by a majority of persons with disabilities at the board and membership levels which undertake service delivery and advocacy and provide the opportunity to develop skills in the negotiation process, organizational abilities, mutual support, information sharing and other vocational skills and opportunities;
- (11) 'Discrimination on the basis of disability' means any distinction, exclusion or restriction on the basis of disability which has the purpose or effect of impairing or nullifying the recognition, enjoyment or exercise, on an equal basis with others, of all human rights and fundamental freedoms in the political, economic, social, cultural, civil or any other field. It includes all forms of discrimination, including denial of reasonable accommodation;
- (12) 'District Disability Rights Tribunal' means the District Disability Rights
 Tribunals established under Section134of this Act;
- (13) 'Establishment' means and includes:
 - a. Departments and Ministries of Government;
 - b. Local authorities and authorities or bodies owned, controlled or aided by the Central or State Government;
 - c. Any statutory or non-statutory body created, owned, financially or administratively controlled or aided by the Central or State Government or any such body performing public or civic functions and includes Government Companies as defined in Section 617 of the Companies Act, 1956;
 - d. Any contractor who has been awarded a public tender;
 - e. Any company, firm, cooperative or other society, association, trust, agency, institution, organization, union, landlord, industry, supplier of goods or

- services, factory or other non-statutory body which is not covered under clause (i) to (iv) and provides education, employment, health care rehabilitation or other services in rural and urban areas;
- f. Any establishment as the Central or State Government may notify in the Official Gazette.

Explanation

The term 'industry' shall have the same meaning as in Section 2 (j) of the Industrial Disputes Act, 1947.

- (14) 'Experiential expert' is a person with disability who has through personal experience, advocacial intervention or service provision acquired expertise in relation to the impairment and the related barriers; or a family care giver who has through personal experience, advocacial intervention or service provision acquired expertise in relation to the impairment and the related barriers;
- (15) Exploitation' occurs when any person having a relationship whether personal, professional or fiduciary or otherwise holding the trust and confidence of the person with disability, knowingly, or by deception or intimidation, obtains or uses or tries to obtain or use her or his funds, assets, or property with an intent to temporarily or permanently deprive such person of their use, benefit, or possession, or to benefit someone other than the person with disability without obtaining her or his consent;
- (16) 'Family care-giver' is a person who is a member of the family of person with disability who provides voluntary care, support and assistance to the person with disability;
- (17) 'Fund' means the National Fund for Persons with Disabilities established under this Act;
- (18) 'Harassment' means unwanted conduct related to any prohibited ground which takes place with the purpose or effect of violating the dignity of a person or of creating an intimidating, humiliating or offensive environment;
- (19) 'Hate Speech' means any kind of expression through any media whether through writing, giving speeches or holding discussions at meetings of groups of individuals with an intention to cause mental trauma, or incite hatred against persons with disabilities or their families;

- (20) 'High Support' means that support, which may be needed by individuals who require ongoing intensive support for activities of daily living; independent and informed decision-making; accessing facilities and participating in all areas of life including education; employment; family and community life; treatment and therapy; recreation and leisure. The need for high support often arises due to significant disability in a single area or multiple or cumulative disabilities in cognition, communication, sensory perceptual processing (not including blindness), emotional regulation, socialization, behaviour and mobility in interaction with attitudinal or environmental barriers:
- (21) 'Indirect Discrimination' occurs when a provision, criterion or practice, even though neutral on the face of it, has the effect of putting persons to whom one or more prohibited grounds apply at a disadvantage compared with others;
- (22) 'Language' includes spoken and signed languages and other forms of non spoken languages;
- (23) 'National Disability Rights Authority' means the National Disability Rights Authority established under Section 94 of this Act;
- (24) 'National Disability Rights Tribunal' means the National Disability Rights Tribunal established under Section 141 of this Act;
- (25) 'Persons with disabilities' are persons with any developmental, intellectual, mental, physical or sensory impairments including those mentioned in Schedule 1¹ of the Act, which are not of a temporary nature, and which in interaction with various barriers may hinder full and effective participation in society on an equal basis with others;
- (26) 'Premises' includes (i) a structure, building, aircraft, vehicle or vessel; and (ii) a place (whether enclosed or built on or not); and (iii) a part of premises (including premises of a kind referred to in sub-clause (i) or (ii);
- (27) 'Prescribed' means prescribed by rules made under this Act;
- (28) 'Prohibited grounds' are the grounds on the basis of which persons with disabilities may be discriminated against, and include the following grounds:
 - a. Disability;
 - b. Perceived disability, whether the perception is accurate or otherwise, or;
 - c. Association of a person with persons to whom one or more prohibited grounds apply, or;

¹As suggested at the Legal Consultation

- d. A combination of disability and any other ground such as religion, race, caste, tribe, place of birth, age, language, sex, gender identity, sexual orientation, pregnancy, maternity, marital status, care giver status, economic status, political or other opinion.
- (29) 'Public building' means a building, irrespective of ownership, which is used and accessed by the public at large; and includes jails, prisons and places of detention;

(30) 'Reasonable Accommodation' means:

- a. where a provision, criterion or practice puts a person to whom one or more prohibited grounds apply at a disadvantage in relation to a relevant matter in comparison with other persons, to take such steps as it is reasonable to have to take to avoid the disadvantage;
- b. where a physical feature puts a person to whom one or more prohibited grounds apply at a disadvantage in relation to a relevant matter in comparison with other persons, to take such steps as it is reasonable to have to take to avoid the disadvantage;
- c. where a person to whom one or more prohibited grounds apply would, but for the provision of an auxiliary aid, be put at a disadvantage in relation to a relevant matter in comparison with other persons, to take such steps as it is reasonable to have to take to provide the auxiliary aid;
- d. where clause (i) or (iii) above relates to the provision of information, the steps which it is reasonable to have to take include steps for ensuring that in the circumstances concerned the information is provided in an accessible format.
- (31) 'Registered Organization' means an association of persons with disabilities or a Disabled Persons' Organisations, association of parents of persons with disabilities, association of persons with disabilities and family members, or a voluntary, non-governmental or charitable organization or trust, society, non-profit company working for persons with disabilities duly registered under relevant Acts for the time being in force;
- (32) 'Self Help Group' means an organization formed by persons with disabilities with joint resources, for gathering information, undertaking advocacy, providing economic support, schemes for income generation, offering community based support services or care through mutual support mechanisms;

- (33) 'Services' means services provided by members of any profession or trade, or provided by any government, local authority or establishment and includes services relating to banking and finance; education; health; insurance; rehabilitation; entertainment, recreation and hospitality; transport or travel; and telecommunications:
- (34) 'State Disability Rights Authority' means the State Disability Rights Authority established under Section 115 of this Act;
- (35) 'State Disability Rights Tribunal' means the State Disability Rights Tribunal established under Section 138 of this Act;
- (36) 'Subject expert' means a person whether with or without disability who has through a formal course of study acquired recognized educational qualifications and has also obtained experience through the practical application of such knowledge;
- (37) 'Support network' refers to a group which supports a persons with disabilities in carrying out his or her life activities. This network may be made of family caregivers, family members, friends, service providers, Disabled Persons' Organizations, self help groups and others who have a personal connection and are in a trusted relationship with the person with disability;
- (38) 'Systemic violation' refers to policies or procedures that violate the rights of persons with disabilities;
- (39) 'Universal design' means the design of products, environments, programmes and services to be usable by all people, to the greatest extent possible, without the need for adaptation or specialized design. "Universal design" shall also apply to assistive devices including advanced technologies for particular groups of persons with disabilities;
- (40) 'Victimization' means subjecting or threatening to subject any of the following persons to any disadvantage:
 - a. one who brings a proceeding under this Act; or
 - b. gives evidence or information in connection with a proceeding under this Act; or
 - c. takes any other steps in connection with or in furtherance of the purpose of this Act; or
 - d. reports (whether express or not) that another person has contravened this Act.

(41) 'Violence' means any random or unprovoked act or threat of such act on any persons with disabilities that harms or injures or endangers the health, safety, life, limb or well-being, whether mental or physical, of the aggrieved person or tends to do so:

3. Guiding Principle for Implementation and Interpretation

Any court, person or authority interpreting this Act shall be guided by the United Nations Convention on the Rights of Persons with Disabilities which has been signed and ratified by India.

PART I: RIGHTS AND ENTITLEMENTS

4. Right to Equality and Non-Discrimination

- (1) All persons with disabilities are equal before the law and are entitled to equal protection and equal benefit of the law without any direct or indirect discrimination;
- (2) No person with disability shall be discriminated on the basis of any of the prohibited grounds; unless it can be shown that the impugned act, provision, criterion, practice, treatment or omission is a proportionate means of achieving a legitimate aim;

Provided that harassment, victimisation or a failure to provide reasonable accommodation may not be justified under any circumstance;

- (3) In no case shall a person with disability be forced or compelled to partly or fully pay any of the costs incurred to provide reasonable accommodation or any of the other obligations imposed in this section;
- **(4)** All persons with disabilities are entitled to equal and effective legal protection from discrimination on any of the prohibited grounds;
- (5) Any measure, intervention, interpretation which has the effect of denying or withdrawing or eliminating the legal capacity of any person with disability shall constitute discrimination;

- (6) Any affirmative action or measure aimed to accelerate or achieve de facto equality of persons with disabilities shall not constitute discrimination;
- (7) In any case upon it being found that an establishment or person has practiced discrimination which such establishment or person has failed to justify, the District Disability Rights Tribunal or the State or National Disability Rights Tribunal as the case may be, shall make declaratory, mandatory, injunctive, compensatory, supervisory or any other suitable orders against such establishment or person to redress the victim for any disadvantage or mental anguish, suffered because of such discrimination and stop any further discrimination against the victim or persons situated in a position similar to the victim;
- (8) The concerned Disability Rights Tribunal shall also require such establishment or person to bear the costs of the proceedings and to pay any interest that may be due.

5. Pro-Active Interventions for Persons with Disabilities with Increased Vulnerability

(1) The appropriate governments shall devise and implement pro-active, dedicated and timely interventions in order to guarantee full enjoyment of all rights and benefits by persons with disabilities with increased vulnerability, on an equal basis with others.

Explanation

Persons with disabilities with increased vulnerability shall mean persons with disabilities who due to the nature of their impairment combined with attitudinal, economic, or locational barriers are subject to familial or social exclusion; and shall include persons with disabilities who are elderly; or confined to their homes²; or are concealed, neglected, abandoned or segregated; or living in institutions; or destitute or homeless.

(2) The interventions referred to in sub section (1) shall be designed to progressively obtain the full inclusion and total participation of persons with disabilities with increased vulnerability in all spheres of life;

² As suggested at Andhra Pradesh State Consultation

(3) The appropriate governments shall consult persons with disabilities and their families to design schemes and programmes for persons with disabilities with increased vulnerability which will pro-actively ensure their meaningful participation in familial and social interactions; assist their access to all services, facilities and information; and ensure on an equal basis the enjoyment of all rights and benefits guaranteed to persons with disabilities;

6. Women and Girls with Disabilities

- (1) All women and girls with disabilities are equal before and under the law and are entitled without any discrimination to the equal protection and equal benefit of the law:
- (2) The appropriate governments and concerned establishments shall take all suitable measures, including devising and implementing gender-specific programmes and schemes, to ensure the full and equal enjoyment of all rights by women and girls with disabilities;
- (3) The appropriate governments and concerned establishments shall accord due recognition to the concerns of women and girls with disabilities by making adequate provision for them in all existing and future women specific schemes and programmes³;
- (4) Every scheme or programme devised for the benefit of persons with disabilities under this Act or under any other law for the time being in force shall make a proportionate allocation for women and girls with disabilities and maintain records in such manner as may be prescribed of the utilization of such allocation;
- (5) Nothing in this section shall prevent the appropriate government or any establishment from taking specific measures for, or providing special entitlements to, women and girls with disabilities to promote full and equal enjoyment of their rights.

³ As suggested at the Delhi, Haryana and Jharkhand State Consultation.

7. Right to Education of Women and Girls with Disabilities

- (1) All appropriate governments and concerned establishments shall take all necessary steps to ensure that no girl or woman with disability shall be excluded from the education system on the grounds of sex or disability;
- (2) The appropriate governments shall take all suitable measures to ensure that all girls and women with disabilities have access to an inclusive education system, without discrimination and on an equal basis with others, at all levels;
- (3) In order to devise the aforesaid suitable measures for girls with disabilities the appropriate governments shall whilst conducting the local survey in Section 36 of this Act, especially collate data on the number of girls with disabilities in the locality; the nature of their impairment and the barriers faced by them to realize the right to education
- (4) Without prejudice to the measures that may be devised pursuant to the local survey the appropriate governments may:
 - a. Provide scholarships for girls with disabilities and special allowance for families of girls with disabilities to meet expenses incurred on their education or vocational training⁴;
 - b. Ensure that all girls with disabilities have access to necessary, adequate and appropriate support for the completion of elementary, secondary and higher education such as accessible transport facilities or other financial incentives to parents and guardians to enable girls with disabilities to attend schools and accessible residential facilities if such facilities are provided for other students;
 - c. Ensure that all schools have appropriate toilet facilities for girls with disabilities.

8. Right to Work and Employment of Women with Disabilities

(1) No woman with disability shall be directly or indirectly discriminate against in recruitment, promotion, or any other related matter arising in the course of or through the length of employment;

⁴ As suggested at the Karnataka State Consultation.

- (2) The appropriate governments shall take all effective and appropriate measures, including formulation of schemes and programmes, to ensure that women with disabilities have access to opportunities for employment, vocational training, micro-credit and self-employment on an equal basis with others;
- (3) Without prejudice to the generality of sub-section (2), such measures shall include:
 - a. according access to working women's hostels for women with disabilities;5
 - b. Mechanisms of support such as personal assistance; mother-hood allowance⁶; extended maternity leave,⁷ flexible working hours and accessible crèches:
 - c. Ensuring that all establishments have appropriate toilet facilities for women with disabilities;
 - d. All committees against sexual harassment shall be under a duty to promote and protect the right of all women with disabilities to a safe working environment; in fulfilment of this duty the committees shall inter-alia: undertake appropriate awareness raising programmes and devise accessible complaint mechanisms.

9. Right of Women and Girls with Disabilities to be Protected against all Forms of Abuse, Violence and Exploitation

- (1) The appropriate governments shall take all necessary measures to protect every woman and girl with disability from all forms of abuse, violence and exploitation including physical, mental, sexual, and emotional on the person of such woman with disability in all settings at all places including, home, care-houses, educational institutions, institutions, workplaces, and any other place where a woman with disability works or resides whether temporarily or permanently;
- **(2)** Without prejudice to the generality of sub-section (1), the appropriate governments shall take measures for:

⁶ As suggested at Karnataka State Consultation

⁵ As suggested at by Orissa State Consultation

⁷As suggested at Maharashtra State Consultation.

- a. Provision of accessible, safe and confidential complaint mechanism to report instances of abuse, violence and exploitation and to address such complaints in a time-bound manner⁸;
- b. Enable the use of short stay and other protection facilities established for women who are victims of abuse, violence and exploitation⁹;
- c. Provision of gender, disability and age sensitive protection services, assistance and support for victims of abuse, violence or exploitation, for physical, cognitive and psychological recovery and development, rehabilitation and social reintegration, in an environment that fosters the health, welfare, self-respect, dignity and autonomy of the person;
- d. Dissemination of information on avoidance, recognition and report of instances of abuse, violence and exploitation.

10. Right of Women and Girls with Disabilities to Health

- (1) The appropriate governments shall take all suitable measures to ensure that all women and girls with disabilities have access to the highest attainable standard of health and health care in accessible environments through accessible procedures with reasonable accommodation without discrimination on the basis of disability or sex;
- (2) The appropriate governments shall take all suitable measures to ensure the full and effective enjoyment of the right to health by women with disabilities, including measures to:
 - a. ensure that all healthcare services, including family planning programmes and services aimed at early identification and intervention, are gender-sensitive, and are available and accessible to women with disabilities, whether in urban or rural areas;
 - b. provide information in accessible formats in relation to all areas of healthcare, including sexual and reproductive health, to women with disabilities.
 - c. The surveys, investigations and research undertaken or caused to be undertaken by appropriate governments and establishments under Section 72

⁸As suggested at Karnataka State Consultation.

⁹ As suggested at Orissa State Consultation

of this Act shall specifically include data or information with respect to women with disabilities 10.

11. Right of Women with Disabilities to Home and Family

- (1) The appropriate governments shall take all suitable measures to eliminate discrimination against women with disabilities in all matters relating to marriage, family, parenthood and relationships, whether the disability is acquired before or after marriage¹¹. Such measures include:
 - a. Providing early and comprehensive information, services and support to women with disabilities and their families to prevent concealment, abandonment, neglect and segregation of women with disabilities in relation to home and family life;
 - b. Providing and making accessible information on reproductive health and family planning to enable women with disabilities to take decisions involving their reproductive rights and family planning freely and responsibly.

12. Access to Justice by Women with Disabilities

The appropriate governments shall take measures to ensure that all police stations, courts, tribunals, or any other body having adjudicatory powers or otherwise connected to the legal system are accessible to women with disabilities.

13. Obligations of the National Commission for Women

The National Commission for Women constituted under section 3 of the National Commission for Women Act 1990 (Act No.20 of 1990) shall in addition to the functions assigned to it under that Act, also perform the following functions, namely:

¹⁰ As suggested at West Bengal State Consultation.

¹¹As suggested at West Bengal State Consultation.

- (1) examine the extent to which the concerns of women with disabilities have been inducted in existing statutes, rules and regulations on women and make appropriate recommendations for such inclusion in consultation with women with disabilities;
- (2) scrutinize whether all proposed legislations for protecting the rights of women duly incorporate the concerns of women with disabilities;
- (3) examine the extent to which the concerns of women with disabilities have been incorporated in existing schemes and programs for women and to recommend requisite modification in consultation with women with disabilities 12.

14. Rights of Children with Disabilities

- (1) The Registrar General appointed under section 3 of the Births and Deaths Registration Act 1969 shall whilst carrying out the superintendence duties of registering births and deaths in the country ensure, including by issuing general directions to all Chief Registrars and other concerned officials, that the registration of births and deaths of all children with disabilities is undertaken in accordance with the procedure specified in the Births and Deaths Registration Act of 1969;
- (2) All children with disabilities have the right to equality before the law and equal treatment of the law and not to be denied any of the guaranteed rights by reason of their age or disability;
- (3) The appropriate governments and concerned establishments shall ensure that all children with disabilities enjoy their human rights and fundamental freedoms on an equal basis with other children and ensure that no child is discriminated on the basis of disability;
- (4) The appropriate governments and concerned establishments shall accord due recognition to concerns of children with disabilities with reasonable accommodation in all existing child specific schemes and programmes;
- (5) All children with disabilities have on an equal basis with all other children a right to freely express their views on all matters affecting them; and all appropriate governments and concerned establishments are under a duty to provide them age and disability appropriate support for the exercise of this right.

¹² As suggested at Manipur State Consultation

15. Right of Children with Disabilities to Health

All appropriate governments and concerned establishments shall ensure screening of new born babies in order to facilitate early intervention and to prevent aggravation of disabilities in children.

16. Right of Children with Disabilities to Leisure, Culture and Sports

- (1) All children with disabilities have a right to play and participate in sports, recreation and cultural activities on an equal basis with other children;
- (2) All appropriate governments and establishments shall provide for disability and age appropriate opportunities for children with disabilities to participate in sports and have access to playgrounds along with other children;
- (3) All appropriate governments and establishments shall ensure that children with disabilities have access to cultural materials in an accessible format and access to cultural activities, performance and services along with other children.

17. Access to Justice for Children with Disabilities

- (1) The National and State Legal Services Authorities shall create panels of Child Rights Defenders to proactively address any discrimination experienced by children with disabilities;
- (2) The National Commission for Protection of Child Rights constituted under section 3 of the National Commission for Protection of Child Rights Act, 2005 (Act no. 4 of 2006) shall whilst performing the functions assigned to it under section 13 of that Act, expressly address the concerns of children with disabilities;

18. Right to Legal Capacity and Equal Recognition before the Law

(1) Notwithstanding anything contained in any other law to the contrary, persons with disabilities enjoy legal capacity on an equal basis with others in all aspects of life and have the right to equal recognition everywhere as persons before the law;

- (2) Any express or implied disqualification on the grounds of disability prescribed in any legislation, rule, notification, order bye-law, regulation, custom or practice which has the effect of depriving any person with disability of legal capacity shall not be legally enforceable from the date of enforcement of this Act;
- (3) Notwithstanding anything contrary contained in any other law, all persons with disabilities have right, on an equal basis with others, to own or inherit property; control their financial affairs; obtain access to bank loans, mortgages and other forms of financial credit, and not to be arbitrarily deprived of their property;
- **(4)** All persons with disabilities have the right to access all arrangements and support necessary for exercising legal capacity in accordance with their will and preferences;
- (5) The legal capacity of a person with disability shall not be questioned or denied, irrespective of the degree and extent of support, by reason of accessing support to exercise legal capacity;
- (6) When a conflict of interest arises between a person providing support and a person with disability in a particular financial, property or other economic transaction, then such supporting person shall abstain from providing support to the person with disability in that transaction.

Explanation

There shall not be a general presumption of conflict of interest just on the basis that the supported person is related to the person with disability by blood, affinity or adoption.

- (7) No person providing support either individually or as part of a network shall exercise undue influence on a person with disability. Such support when provided by an individual or a network shall be so provided that it respects the autonomy, dignity and privacy of persons with disabilities;
- **(8)** A person with disability may alter, modify or dismantle any support arrangement and substitute it with another.

Provided that such alteration, modification or dismantling shall be prospective in nature and shall not nullify any third party transactions entered into by the person with disability with the aforesaid support arrangement.

19. Replacement of Plenary Guardianship with Limited Guardianship

(1) After the commencement of this Act, any provision in any legislation, rule, regulation or practice which prescribes for the establishment of plenary guardianship shall be hereinafter deemed to be establishing a system of limited guardianship;

Explanation

Plenary Guardianship¹³ is a system whereby subsequent to a finding of incapacity a guardian substitutes for the person with disability as the person before the law and takes all legally binding decisions for him or her. The decisions of the person with disability have no binding force in law during the subsistence of the guardianship. The guardian is under no legal obligation to consult with the person with disability or determine his or her will or preference whilst taking decisions for him or her.

- (2) Subsequent to the enforcement of this Act all plenary guardians shall operate as limited quardians;
- (3) All limited guardians shall act in close consultation with the person with disability to arrive at legally binding decisions.

Explanation

A limited guardianship is a system of joint decision making which operates on mutual understanding and trust between the guardian and the person with disability.

20. Duty to Provide Support

(1) The appropriate governments shall establish or designate one or more authorities to mobilize the community and create social networks to support persons with disabilities in the exercise of their legal capacity. Such authorities shall:

¹³ As suggested at the Delhi, Harayana, Karnataka, Manipur and West Bengal State Consultations

- a. take immediate steps to put in place suitable support measures for the exercise of legal capacity by persons with disabilities living in institutions and persons with disabilities who need to have high support;
- take suitable steps including, where appropriate, mediation proceedings, in order to assist persons with disabilities who have exited from plenary guardianship to set up where required by such persons with disabilities suitable support arrangements other than limited guardianship for the exercise of their legal capacity;
- c. take steps to review how the system of deemed limited guardianship is operating for persons with disabilities including those living in institutions and to assist such persons with disabilities in establishing suitable support arrangements to exercise their legal capacity;
- (2) The review process referred to in sub clause (1) (c) is facilitative in nature and no person with disability can be denied legal capacity or refused the benefit of limited guardianship arrangements due to the delay or non occurrence of the review.

21. Denial of Legal Capacity

- (1) Any act, order or proceedings which has the effect of denying the legal capacity of a person with disability in any matter or which questions the legal capacity of a person with disability on the grounds of disability shall be void;
- (2) Upon a person with disability being denied legal capacity, the aggrieved person with disability may file a complaint before the District Disability Rights Tribunal in addition to other statutory or constitutional remedies.

22. Right to Life and Living

(1) The appropriate governments shall take all measures to ensure that persons with disabilities enjoy the right to life guaranteed by Article 21 of the Constitution of India on an equal basis with others;

- (2) These measures shall be in furtherance of the right to live with dignity for persons with disabilities and to ensure that full value is accorded to the lives of persons with disabilities;
- (3) Any procedure or practice which diminishes the value of the life of persons with disabilities is hereby prohibited;
- (4) In order to assist in the enforcement of the prohibition specified in sub-section (3) the National and State Disability Rights Authority shall through social investigations, public hearings or research studies collate information on such procedures and practices and forward the same to the Central or State Government as the case may be

Provided that if the State Disability Rights Authority is enquiring into a particular practice or procedure then the National Disability Rights Authority shall refrain from doing so.

(5) Any person who through an act or omission endangers the life of a person with disability or damages or injures or interferes with the use of any limb, or faculty of a person with disability shall be punishable under Section 151 of this Act.

23. Right to Liberty

- (1) The appropriate governments and establishments shall take all measures including through the provision for reasonable accommodation and support to ensure that persons with disabilities enjoy the right to personal liberty guaranteed by Article 21 of the Constitution of India on an equal basis with others;
- (2) These measures shall be so designed as to further the right to a non-coercive, non-restrictive and supportive environment for persons with disabilities which respect their sense of space, safety and security;
- (3) No person shall be deprived of personal liberty on the grounds of disability;
- (4) The liberty of persons with disabilities including prisoners with disabilities shall not be deprived except by a procedure which is fair just and reasonable. Such procedure shall require the appropriate government and concerned authorities to provide reasonable accommodation to persons with disabilities including prisoners with disabilities;
- **(5)** Right to personal liberty for persons with disabilities includes:
 - a. freedom from restrictions or encroachments on person, whether those restrictions or encroachments are directly imposed or indirectly arise;

- b. freedom from non-consensual civil commitment
- (6) In furtherance of the right specified in sub section (5):
 - a. the care and treatment of all persons with disabilities shall be so organized that it advances personal liberty and consensual care; whilst restricting and phasing out procedures of compulsory care and non consensual treatment;
 - b. All appropriate governments shall subsequent to the commencement of this Act order a review of all cases of persons placed under non consensual civil commitment to determine the circumstances which necessitated compulsory confinement and the initiatives required to promote community living and consensual care of persons with disabilities.
- (7) In order to protect and promote the freedoms guaranteed in sub sections (5) and (6) of this section, the appropriate governments and establishments shall, in consultation with the State and National Disability Rights Authority, in partnership with non-governmental organizations or otherwise launch programs by which inter alia:
 - a. values of peace, love, non-violence and respect for difference and diversity in schools is promoted;
 - b. safe havens, time out sites and respite places are available to diffuse situations of conflict and violence;
 - c. the importance of difference is appreciated and skills of dealing with diversity developed;
 - d. Community leaders trained who can address conflict situations through conciliation and arbitration between the community and within the family so as to protect the liberty of all persons with disabilities and especially women, children and the elderly.

24. Right to Live in the Community

- (1) All persons with disabilities shall have the right to live in the community in furtherance of their right to have a meaningful and full life on an equal basis with others:
- (2) The appropriate governments shall launch suitable schemes and programs including community based rehabilitation and such other initiatives in order to realize the right recognized in sub section (1). Such schemes and programs shall:

- a. provide choices to persons with disabilities in the place and manner of living and shall not direct or order persons with disabilities to live in any particular arrangement;
- b. accord due recognition to the age and gender needs of persons with disabilities;
- c. be accompanied with such support as may required by persons with disabilities to realize their choice.
- (3) The appropriate governments shall accord due consideration to gender and age needs whilst establishing independent and community living environments;
- (4) The appropriate governments shall further ensure that such independent living environments will be accompanied with access to such support as may be necessary to persons with disabilities to fully realize their choice in the matter of living arrangements, on an equal basis with others.

25. Right to Integrity

- (1) Every person with disability has a right to respect for his or her physical and mental integrity on an equal basis with others;
- (2) All appropriate governments and establishments shall ensure that the identity of all persons with disabilities is respected and they are able to function safely as such persons;
- (3) The respect mandated by sub-sections (1) and (2) shall in no way be influenced by the living arrangements of the person with disabilities. The right of homeless persons with disabilities and persons with disabilities in institutions are to be respected on an equal basis with others;
- (4) Any communication, address or greeting which diminishes the identity of persons with disabilities is hereby prohibited;
- (5) No person with disability shall be subjected to any research protocol without his or her free and informed consent. Such consent must be obtained through accessible modes means and formats of communication:
- (6) All research projects which relate to persons with disabilities or involve them in any manner shall be undertaken only subsequent to obtaining clearance from an

Ethics Committee which either has a member of a Disabled Persons' Organizations or a Disability Rights Defender as member.

26. Protection from Abuse, Violence and Exploitation

- (1) No person shall act or abstain from acting where such action or inaction causes or is likely to cause abuse, violence and exploitation to persons with disabilities at any place including, home, care-houses, institutions, educational institutions, workplaces, or any place where such persons with disabilities reside, co-habit, work or in other manner inhabits whether temporarily or permanently;
- (2) The appropriate governments shall take all appropriate administrative, social, educational and other measures to protect persons with disabilities, both within and outside the home, from all forms of abuse, violence and exploitation;
- (3) Any person, or registered organization who has reason to believe that an act of abuse, violence and exploitation has been, or is being, or is likely to be committed against any person with disability, may give information about it to the District Disability Rights Tribunal in whose jurisdiction such incident occurs is or is likely to occur, who on the receipt of such information shall take immediate steps to stop it or prevent its occurrence as the case may be or pass any order it deems fit for the protection of such person with disability including an order:
 - a. to rescue the victim of such act, authorizing the police or any reliable organization working with the person with disability to provide for the safe custody, or rehabilitation of such person with disability, or both, as the case may be;
 - b. for providing protective custody of the person with disability if the person with disability so desires it;
 - c. to provide maintenance to such person with disability.
- (4) No civil or criminal liability shall be incurred by any person who in good faith furnishes information for the purpose of sub section (3);
- **(5)** Any police officer who receives a complaint or comes to know otherwise of abuse, violence or exploitation towards any person with disability shall inform the aggrieved person of:

- a. the right to apply for protection under subsection (3).
- b. the particulars of the nearest organization or institution working for the rehabilitation of persons with disabilities who have been subject to abuse, violence or exploitation;
- c. the particulars of the nearest office(s) of the authorities mentioned under this Act, which shall assist the complainant;
- d. of the right of persons with disabilities to free legal services under the Legal Services Authorities Act, 1987 (Act No. 39 of 1987) and any other services offered by the National Legal Services Authority or the State Legal Services Authority for the benefit of persons with disabilities;
- e. of the right to file a complaint under the relevant provisions of the Indian Penal Code or any other legislation dealing with such crimes.

Provided that nothing in this Act shall be construed in any manner as to relieve a police officer from his duty to proceed in accordance with law upon receipt of information as to the commission of a cognizable offence.

- (6) If the District Disability Rights Tribunal finds that the alleged act or behavior would also be an offence under the Indian Penal Code, 1860 (Act No. 40 of 1860) or under any other law imposing criminal sanctions on such acts, it may direct the Disability Rights Defender to file the complaint before the judicial or metropolitan magistrate, as the case may be, having jurisdiction in the matter and the Disability Rights Defender shall act forthwith in accordance with such direction;
- (7) The appropriate governments shall take all appropriate measures to prevent all forms of abuse, violence and exploitation by, inter alia, providing information and raising awareness on:
 - a. identification and recognition of the incidents of abuse, violence and exploitation;
 - b. the legal remedies available against such incidents;
 - c. steps to be taken for avoiding such incidents;
 - d. steps to be taken for reporting such incidents;
 - e. steps required for the rescue, protection and rehabilitation of persons with disabilities who have been the victims of such incidents.

27. Protection and Safety of Persons with Disabilities in Situations of Risk

- (1) All persons with disabilities shall have the right to suitable protection and safety in situations of risk, including situations of armed conflict, humanitarian emergencies and natural disasters:
- (2) In furtherance of the right recognized in sub-section (1), the appropriate governments and local authorities shall take effective and appropriate measures to enable the protection and safety of persons with disabilities in these situations on an equal basis with others;
- (3) Without prejudice to the generality of the obligation in sub-section (2), the appropriate governments shall formulate or alter as need be, all existing and proposed plans and interventions, in consultation with the National and State Disability Rights Authority as the case may be, for the inclusion of persons with disabilities;
- (4) The National and State Disaster Management Authority shall take appropriate measures to ensure inclusion of persons with disabilities in every scheme, program, mission as well as its 'disaster management activities' as defined under S 2(e)of the Disaster Management Act, 2005(Act No 53 of 2005) during the occurrence of any 'disaster' as defined under S. 2(d) of the (Disaster Management Act, 2005 for the safety and protection of persons with disabilities;
- (5) The National and State Disaster Management Authority shall prepare and maintain records with contact of persons with disabilities in an area and take suitable measures to inform such persons of any situations of risk so as to enhance disaster preparedness¹⁴;
- (6) The appropriate governments shall take suitable measures to ensure the safety and protection and to offer immediate relief and rehabilitation to victims with disabilities in situations of communal violence or internal disturbance;
- (7) If due to the activities of the government in a conflict zone, any person acquires an impairment; or a person with disability acquires further impairments or an existing impairment is further aggravated; the Appropriate Government shall provide compensation and reimbursement of medical expenses within the time

¹⁴ As Suggested at Maharashtra State Consultation

- and in accordance with the scales as may be prescribed. Such scales shall accord due weightage to considerations of gender and age¹⁵;
- (8) The Armed Forces shall, in all situations of armed conflict, take appropriate measures to ensure the safety and protection of persons with disabilities in the light of the obligations under International Human Rights law and International Humanitarian law;
- (9) All authorities engaged in reconstruction activity subsequent to any situation of armed conflict, humanitarian emergencies or natural disasters shall undertake such activity, in consultation with the National Disability Rights Authority, in accordance with the accessibility requirements of persons with disabilities.
- (10) The appropriate governments and concerned authorities shall in consultation with the National and State Disability Rights Authority:
 - a. undertake research and development on issues relating to protection, relief, rescue or safety of persons with disabilities in situations of risk;
 - b. provide training to personnel dealing with persons with disabilities in situations of risk, including security forces in conflict areas¹⁶, so as to enable them to offer effective support in these situations and sensitize them to the needs and requirements of persons with disabilities; and
 - c. inform and train persons with disabilities on appropriate responses to the interventions of the authorities in situations of risk.

28. Right to Home and Family

- (1) All persons with disabilities especially children and elderly shall have the right to home and family on an equal basis with others;
- (2) Every person with disability shall have the right to marry a person of his or her choice without discrimination on basis of disability;
- (3) No person with disability shall be denied guardianship or custody of his or her child on the grounds of disability;
- (4) All persons with disabilities are entitled to adopt a child or to give a child in adoption on an equal basis with others;

¹⁵As Suggested at the J&K State Consultation.

¹⁶ As Suggested at the Manipur State Consultation

- (5) No child with disability shall be separated from his or her parents on grounds of disability except on an order of competent Court if required in the best interest of the child;
- (6) The competent Court shall make every effort to place the child with disability within the extended family and failing that within the community in a family setting.

Explanation

Family means a group of people related by blood, marriage or adoption to the person with disability

29. Additional Provision for Disability-Related Needs in Maintenance and Alimony

- (1) Where maintenance or alimony is awarded to a woman with disability under the relevant applicable law, the court shall consider the disability related needs of the woman while determining the amount of such maintenance or alimony;
- (2) While awarding maintenance or child support for a child with disability under the relevant applicable law the Court shall consider the disability related needs of the child while determining the amount of such maintenance or alimony to be awarded.¹⁷

30. Reproductive Rights of Persons with Disabilities

- (1) All persons with disabilities, particularly women and children with disabilities, shall have the right to retain their fertility;
- (2) The appropriate government shall ensure that persons with disabilities have access to information regarding family and reproductive planning on an equal basis with others;
- (3) No person with disability shall be subject to any medical procedure which leads to or could lead to infertility without their free and informed consent;
- (4) Any contravention of this provision shall be penalized under Section153.

¹⁷ As suggested at the Legal Consultation

31. Freedom of Speech and Expression

- (1) All persons with disabilities have the freedom to express their thoughts and opinions on an equal basis with others;
- (2) No person, establishment, or government shall censor or curtail the thought, speech, expression or communication of a person with disability on basis of disability; any such censorship or curtailment shall be discrimination on the basis of disability;
- (3) The expression of dissenting opinions by persons with disabilities enjoys the same constitutional respect as other voices of dissent; any curtailment or gagging of such dissent shall be discrimination on the basis of disability;
- **(4)** Any person practicing hate speech will be punishable under Section 154 of the Act.

32. Right to Political Participation

- (1) Notwithstanding anything contained in the Representation of the People Act (No. 43 of 1950), the Representation of the People Act (No. 43 of 1951) or any other law for the time being in force, every person with disability who fulfills all the eligibility requirements shall be entitled to be registered as a voter and shall not be held disqualified to exercise his or her right to vote on the ground of disability irrespective of any stipulation to the contrary in any law for the time being in force¹⁸;
- (2) Notwithstanding anything contained in the law for the time being in force, any person with disability who is unable to cast vote in person due to his or her disability or because of admission in any establishment maintained wholly or mainly for the reception and treatment of persons with disabilities at the time the poll is taken shall be entitled to vote by postal ballot and provisions shall be made, by rules made under the Representation of the People Act (No. 43 of 1951) to enable such voting;

¹⁸ As suggested at the Uttarakhand State Consultation.

- (3) The Election Commission shall ensure that all polling stations are accessible to persons with disabilities and that all materials related to the electoral process are easily understandable by and accessible to persons with disabilities;
- (4) Without prejudice to the generality of the provisions contained in sub–section (3), the measures undertaken by the Election Commission in pursuance of that sub–section shall include:
 - a. the construction and availability of ramps at all polling booths;
 - b. separate queues for persons with disabilities at all polling booths with clear pictorial signs;
 - c. the availability of ballot papers and/ or electronic voting machines with candidates' information available in Braille and other accessible formats;
 - d. the fitting of audio devices to electronic voting machines;¹⁹
 - e. training programs to sensitise polling officers about the special requirements of persons with disabilities.
- (5) If the presiding officer is satisfied that, due to disability, a person with disability is unable to recognise the symbols or to record vote without assistance, the presiding officer shall permit the elector to take a companion of not less than eighteen years of age to the voting compartment for recording/ casting the vote;
- **(6)** Every person with disability is entitled to form and be a member of any organization or association in accordance to choice.

33. Access to Justice

- (1) All persons with disabilities and Disabled Persons' Organizations shall, in their individual or representative capacity, as the case may be, have the right to access any court; tribunal; authority; commission; or any other body having judicial or quasi judicial or investigative powers on an equal basis with others;
- (2) The authorities in subsection (1) shall ensure that persons with disabilities are able to exercise the right to access these authorities without discrimination on the basis of disability and provide reasonable accommodation whenever required;

 $^{^{\}rm 19}$ As suggested at the Karnataka State Consultation. .

- (3) The right of access to justice in subsection (1) includes, initiating a legal proceeding, participate in any legal proceeding as a party, complainant, accused, advocate, witness, expert, or amicus curiae or in any other capacity relevant for and to assist in adjudication of any dispute. No person with disability shall be denied the right to participate on the grounds of disability;
- (4) Every person with disability shall be entitled to use any means of communication along with support which such person desires and finds appropriate to participate in legal proceedings;
- (5) All communication under sub section (3) shall be deemed to be oral communication for all purposes including under the Code of Criminal Procedure, 1973 (Act No. 2 of 1974) and the Indian Evidence Act, 1872 (Act No. 1 of 1872);
- (6) All the authorities mentioned in subsection (1) shall evaluate the testimony, opinion or argument given by a person with disability on an equal basis with others and ensure that there is no discrimination whether directly or indirectly on the basis of disability;
- (7) National Legal Services Authority and the State Legal Services Authority of the respective State shall make provisions including reasonable accommodations to ensure that persons with disabilities have access to any scheme, programme, facility or service offered by them on an equal basis with others.
- (8) The National Legal Service Authority and the State Legal Service Authority of all the states shall formulate such schemes, procedures, facilities, support systems and arrangements to enable persons with disabilities to effectively exercise their right under sub-section (1), (2) and (3) of this section;
- (9) Every appropriate Government shall take immediate steps to ensure that all police stations are accessible to all persons with disabilities and the necessary equipments, support or arrangements are made available to enable persons with disabilities to file a First Information Report, join investigation, and get their statement recorded before the police.
- (10) All the concerned agencies or authorities including the courts and tribunals at different levels shall take steps to make their premises, offices and infrastructure accessible to persons with disabilities;
- (11) The authorities in sub-section (1), shall take immediate steps to:
 - a. ensure that all their documents are in accessible formats:

- b. ensure that filing departments, registry or any other office of records are supplied with necessary equipments to enable filing, storing and referring to the documents and evidence in accessible formats;
- c. make available all the necessary facilities and equipments to facilitate recording of testimonies, arguments or opinion given by persons with disabilities in their preferred language and means of communication;
- (12) All Judicial Academies; Police Academies and Training Organizations for Prison Officials and Legal Service Authorities shall undertake regular training programmes on disability rights as a part of their induction and continuing education programmes.
- (13) All authorities mentioned in subsection (1) shall take steps to ensure, that matters involving the rights and interests of persons with disabilities are disposed of expeditiously.

34. Definitions for Right to Education

- (1) For the provisions relating to this right unless the context otherwise requires:
 - a. 'child' means a child with disability who is below the age of eighteen years;
 - b. 'neighbourhood school' is any educational institution providing elementary education within a radius of one kilometer or secondary education within a radius of three kilometers which is suitably equipped to provide education to children with disabilities and shall also include schools that are exclusively or dominantly catering to children with disabilities;

Provided that in areas which have a difficult terrain, or are prone to landslides, floods, or any other barrier the school may be located outside this radius so as to avoid these dangers. And would include home or hostel based schools if no such site is available near the home of the child.

- c. 'secondary education' means education from the ninth class to the twelfth class;
- d. 'special educator' includes teachers, trainers, resource persons trained to address the general and particular learning needs of children with disabilities;

- e. 'special school' means a school established and run mainly²⁰ for students with disabilities:
- f. 'support' includes:
 - a. educational instruction, through appropriate means and methods of communication by educators who have the requisite qualifications and training;
 - b. the free supply of books, preparatory materials for entrance examinations and study materials in appropriate means and methods of communication;
 - c. the free supply of assistive devices required by persons with disabilities for the completion of elementary, secondary or higher education;
 - d. the provision of scribes or other technological support as required;
 - e. transport facilities for children or in the alternative financial incentives to parents and guardians to enable children to attend school;
 - f. scholarships for students with disabilities with preference to women and girls with disabilities;
 - g. extra time to complete tests and examinations as appropriate;
 - h. spelling concessions in written assignments, tests and examinations;
 - exemption from second and third language courses if so desired by a child's parents or guardian and deemed necessary by the school administration;
 - j. suitable technological equipment which assists in the learning of all disciplines including but not limited to mathematics, geography, economics and science courses;
 - k. necessary physical adaptations including classroom, sports facilities, libraries and laboratories²¹;
 - I. any other appropriate aid required for the completion of elementary, secondary and higher education.
 - m. alternative questions to visual questions in examinations or assessments papers²².
- (2) The definition of 'capitation fee' "elementary education" "guardian" " parent" and " screening procedure" shall be as provided in the Right of Children to Free and Compulsory Education Act, 2009 (Act No 35 of 2009)

²⁰ As suggested at the Mizoram State Consultation

²¹As suggested at the Uttarakhand consultation

²² As suggested at the MP State Consultations

(3) The provisions of sections 5, 10, 13, 14 of the Right of Children to Free and Compulsory Education Act, 2009 (Act No 35 of 2009) shall mutatis mutandis apply.

35. Right to Education

- (1) All persons with disabilities have a right to education to enable the full development of their human potential, sense of dignity and self-worth; to develop their personality, talents and creativity, mental and physical abilities to their fullest potential; and to enable their effective participation in an inclusive²³ society;
- (2) No persons with disabilities shall be excluded from the education system on the basis of disability, and the appropriate government shall ensure that all persons with disabilities, especially girls and women with disabilities, have access to education, without discrimination and on an equal basis with others, at all levels.

36. Surveying for Children with Disabilities

- (1) Subsequent to the enactment of this Act and before its enforcement the appropriate government shall conduct local level surveys throughout the country in order to determine the physical presence of children with disabilities and to devise educational plans for the establishment of suitable neighbourhood schools resource centers or special schools as the case may be;
- (2) The appropriate governments shall conduct such surveys every five years

37. Establishment of Neighbourhood Schools, Resource Centers and Special Schools

(1) For carrying out the purposes of this Act, the appropriate governments shall:

²³ As suggested at the Karnataka State Consultation.

- a. establish neighbourhood schools and special schools within such area or limits of a neighbourhood as may be prescribed and all such schools shall be equipped to provide education to all persons with disabilities;
- b. resource centers equipped with requisite personnel and equipment which shall support students and teachers in the pursuit of their education in neighbourhood schools;
- c. develop and enforce standards for training educators to teach children with disabilities;
- d. provide infrastructure including school buildings, educators who have the requisite qualifications and training to teach children with disabilities; learning materials and any other materials required to support a child with disability in the completion of his or her elementary and secondary education;
- e. Ensure and monitor the admission, attendance and completion of elementary and secondary education of every child with disability;
- f. Ensure the availability of a sufficient number of educators who have the requisite qualifications and training to teach persons with disabilities;
- g. Provide training facilities for teachers such that they are trained to cater to the needs of children with disabilities;
- h. Establish an adequate number of teacher training institutions and assist the National Institutes and other voluntary organizations to develop teacher training programs specializing in training special educators so that the requisite special educators are available for students with disabilities;
- Develop and establish an inclusive undergraduate and postgraduate degree in education which trains all teachers to cater to the needs of a child with disability in an inclusive classroom; and
- j. Initiate or cause to be initiated research by official and non governmental agencies for the purpose of designing and developing new assistive devices, teaching aids, special teaching materials, or such other items as are necessary to give a child or person with disability the support required to complete his or her education.

38. Right to Free Childhood Care and Preschool Education

Every child below the age of six years has a right to free childhood care and preschool education.

39. Entitlements of Children with Disabilities

- (1) Every child shall have a right to free and compulsory education in an appropriate neighbourhood school or special school, as chosen by the parents or guardian, between the age of six and eighteen years or until the completion of secondary education whichever is later:
- (2) No child shall be liable to pay any kind of fee, charge or expenses towards any support or otherwise which may prevent him or her from pursuing for completing both elementary and secondary education.

40. Reasonable Accommodations in Education

- (1) The appropriate governments and establishments shall ensure that reasonable accommodation of the individual's requirements is provided at all levels of the education system, taking into consideration both gender and age²⁴ specific needs;
- (2) The appropriate governments and establishments shall ensure that persons with disabilities receive individualised support, within the education system, to facilitate their effective education provided in environments that maximize academic and social development, consistent with the goal of full inclusion;
- (3) In order to ensure appropriate quality²⁵ education for persons with disabilities, the appropriate governments and establishments shall take measures, including:
 - a. facilitating the learning of Braille, alternative script, augmentative and alternative modes, means and formats of communication and orientation and mobility skills, and facilitating peer support and mentoring;
 - b. facilitating the learning of sign language and the promotion of the linguistic identity of the deaf community;

²⁴ As suggested at the Uttarakhand State consultation

²⁵ As suggested at the Uttarakhand State consultation

c. employing teachers, including teachers with disabilities, who are qualified in sign language or Braille, and to train professionals and staff who work at all levels of education.

41. Student-Teacher Ratio for Children with Disabilities

Every school shall maintain such suitable pupil – teacher ratio between students with disabilities and educators who have the requisite qualifications and training as may be prescribed. In determining the ratio the appropriate authority shall ensure that the ratio should be so specified that it enables persons with disabilities to reach the goals of education specified in this chapter.

42. Right to School Admission

If a child has not been admitted to a school at the stipulated age or was admitted but was unable to complete his education due to his or her disability then such child shall be admitted to a class suitable to his or her age.

Provided that, where a child is directly admitted to an age appropriate class such child has a right to suitable training in order to enable effective participation.

Provided further, that the aforementioned training will be imparted through means of communication and in a language which the child understands by educators equipped and trained to perform such training.

43. Right to Support and Accessibility

- (1) Every child has a right to receive necessary, adequate and appropriate support for the completion of elementary and secondary education;
- (2) Every child has a right to access all school buildings, sports recreation and library facilities on an equal basis with others and it shall be the duty of appropriate

governments and establishments to render the aforementioned facilities accessible to children with disabilities.

44. No Detention

No child, who is admitted in a school, shall be held back in a class or expelled from school prior to the completion of secondary education.

45. Impermissible Disciplinary Measures

- (1) No child shall be subjected to physical punishment or mental harassment or in any manner discriminated on the basis of disability;
- **(2)** The denial of support or reasonable accommodation as a disciplinary measure is prohibited.

46. Parents of Children with Disabilities in School Management Committees

- (1) Parents or guardians of children with disabilities shall be members of the School Management Committees established under Section 21 of the Right of Children to Free and Compulsory Education Act, 2009 (Act No 35 of 2009);
- (2) The percentage of parents or guardians of students with disabilities on aforementioned committees shall be proportionate to the ratio of such students in the school;

Provided that in no case shall there be less than two such parents or guardians on the Committee.

47. Disability Related Needs in School Development Plans

(1) All School Development Plans prepared under sub – section (1) of Section 22 of the Right of Children to Free and Compulsory Education Act, 2009 shall specify the resource requirements needed to:

- b. Provide support to children as mentioned in Section 43 and
- c. fulfill education entitlements of children with disabilities.

48. Education Reform Commission for Advancement of Disability Rights Perspective

There shall be established an Education Reform Commission for such period of time, not less than three years, as the central government may by notification provide.

- (1) The Commission shall, to the maximum extent possible, involve an effective participation of all stakeholders in the process of formulation, implementation and monitoring of the curriculum and related programmes and policies, including disabled and non-disabled children, teachers and parents;
- (2) The terms of reference of the Education Reform Commission shall be:
 - a. To review the existing curriculum being adopted in schools from the standpoint of persons with disabilities and their lived experiences;
 - b. To develop an inclusive curriculum based on the principles of nondiscrimination and appreciation of diversity and tolerance;
 - c. To make recommendations on the pedagogical methodology to be adopted for the teaching to and learning by persons with disabilities in the creation of such inclusive curriculum:
 - d. To suggest measures for the adoption and integration of the inclusive curriculum in mainstream education and to monitor its progress²⁶;
 - e. Make such other recommendations as the Commission may consider necessary.
- (3) The Commission may initiate studies and analysis or take any other measures that may be necessary for performing any function in relation to its terms of reference.

49. Right to Higher Education

²⁶ As suggested at the Maharashtra State Consultation.

- (1) No person with disability particularly women with disabilities shall be denied admission in a higher education institution on the grounds of his or her disability; Provided that, if a person with disability is unable to seek admission to a higher educational institution at or before the age prescribed for such admission then he or she shall be given admission in such institution if he or she fulfills all other eligibility requirements.
 - (2) All appropriate governments and educational authorities may if they deem fit to promote equality of opportunity relax according to prescribed procedure the minimum qualification criteria required to be obtained by persons with disabilities who seek admission to a higher education institution.

50. Reservation in Higher Educational Institutions

All institutions of higher education shall reserve not less than six percent of total seats in each course for persons with disabilities.

Provided that persons with disabilities shall not be prevented from competing for seats which are not reserved for them.

51. Support in Higher Education

- (1) Every person with disability appearing for an entrance examination for any higher educational institution has a right to adequate, necessary and appropriate support for the purposes of such examination;
- (2) Every person with disability has the right to receive necessary, adequate and gender sensitive support for the completion of higher education and for any other extra curricular and co curricular activities;
- (3) All universities and other institutions of higher learning shall establish equal opportunity offices to ensure the dedicated provision of such support.

52. Qualifications of Educators

- (1) Every school and higher educational institution shall have a staff of educators who have the requisite qualifications and training to cater to the needs of students with disabilities;
- (2) Teachers in all types of schools should be imparted orientation and training to understand specific needs of children with disabilities.

53. No Capitation Fee in Higher Education

No institution of higher education shall charge or collect any capitation fee whilst admitting a person with disability.

54. Leisure Culture Sport in Schools and Higher Educational Institutions

- (1) All schools shall ensure that children with disabilities shall have the right to participate in sporting, recreational and leisure activities on an equal basis with others:
- (2) All higher educational institution shall ensure that persons with disabilities have the right to participate in sporting, recreational and leisure activities on an equal basis with others.

55. Adult Education for Persons with Disabilities

All appropriate governments and establishments shall ensure participation of persons with disabilities in adult education and continuing education programmes on an equal basis with others

56. Non Discrimination in Employment

(1) No establishment shall directly or indirectly discriminate against any person with disability in any matter relating to employment including but not limited to

- recruitment, promotion and other related issues arising in the course of or through the length of employment in any establishment²⁷.
- (2) No establishment shall dispense with, or reduce in rank, an employee who acquires a disability during service, such employee may if required by the nature of disability, be shifted to another post with the same pay scale and service benefits.

Provided further that if it is not possible to adjust the employee against any post, then such employee may be kept on a supernumerary post until a suitable post is available or the age of superannuation whichever is earlier

(3) The protection accorded in sub section (2) shall also be extended to persons employed with the defense forces;

Provided that this sub section in no way precludes the defense establishment to formulate a more beneficial employment retention and rehabilitation scheme for persons employed with the defense forces.

(4) Every establishment shall facilitate reasonable accommodation of persons with disabilities by taking adequate measures to guarantee that persons with disabilities are not disadvantaged in any manner at any stage of employment.

Explanation

'Adequate Measures' include, but are not limited to the provision of necessary aids and equipment, adequate healthcare facilities, necessary physical changes in buildings to ensure accessibility at workplaces, flexible work timings, continuous monitoring with regard to necessary support, or any arrangements or facilities created for equality with regard to competitive public service examinations and other such service related tests.

- (5) Any person with disability, if eligible for any post which is sought to be filled, shall have the right to appear for selection and hold the post if selected;
- (6) An establishment shall not ordinarily post and transfer a person with disability in a place other than his or her native place or within the vicinity²⁸ of such place unless such transfer becomes necessary due to exigencies of the job and expertise possessed by the person with disabilities;

²⁷ As suggested at the Uttarakhand State Consultation

²⁸ As suggested at the Uttarakhand State Consultation

- (7) The appropriate governments may frame such rules and regulations as may be necessary from time to time for the purposes of achieving the objectives outlined above;
- (8) Every establishment undertaking an exercise of retrenchment or declaring its staff surplus shall as far be not include persons with disabilities in such exercise or process. In the event of persons with disabilities being included in such exercise or process, enhanced benefits shall be payable to them.

57. Reservations

- (1) All establishments shall reserve not less than seven percent of all posts and in promotions for persons with disabilities in accordance with the following banding of disabilities, with each band being entitled to 1%;
 - a. Persons with blindness;
 - b. Persons with hearing impairment and speech impairment;
 - c. Persons with locomotor disability and leprosy cured;
 - d. Persons with cerebral palsy and muscular dystrophy;
 - e. Persons with autism, intellectual disability and mental illness;
 - f. Persons with multiple disabilities, deaf-blindness and multiple sclerosis.
 - g. Persons with Low vision and persons who are hard of hearing

Provided that posts identified under Section 32 of the Persons with Disabilities (Equal Opportunities, Protection of Rights and Full Participation) Act of 1995 (Act No 1 of 1996) shall operate as guidelines for implementing this reservation.

- (2) The percentage stated can be renewed and the classification altered upon review once every 3 years;
- (3) If sufficient number of qualified persons with disabilities are not available in a particular year, then the reservation may be carried forward to the subsequent year, and if in the subsequent recruitment year also a suitable person with disability is not available, then the post may be first filled by interchange among the categories, and if the vacancy is still not filled then the establishment shall identify, train and then induct persons with disabilities on the post. In no case shall a post to which a person with disability is entitled to be appointed be allowed to lapse;

(4) The entitlement outlined in sub section (2) and the process in sub section (3) shall also extend to schemes guaranteeing employment, or the creation of posts for employment, by way of any legislation, notification or rule formulated by the appropriate government, including but not limited to the Mahatma Gandhi National Rural Employment Guarantee Act, 2005.

58. Equal Opportunity Policies

- (1) Subject to the requirements of Section 8 of this Act, within six months of the commencement of this legislation, all establishments shall put in place an Equal Opportunity Policy detailing measures and commitments initiated by the establishment in pursuance of the provisions of this Chapter and any Rules thereof;
- (2) An Equal Opportunity Policy shall:
 - a. delineate measures taken in order to comply with the provisions of the Act;
 - b. provide strategies to increase employment opportunities with specific attention to all schemes and reasonable accommodation measures;
 - c. specifically detail measures taken and strategies employed to reasonably accommodate and increase employment opportunities for women with disabilities.
- (3) The State Disability Rights Authority may, if it deems fit, devise the form and structure of Equal Opportunities Policies;
- (4) Once formulated, a copy of the Equal Opportunity Policy shall be deposited and registered with the State Disability Rights Tribunal.

59. Maintenance of Records

(1) Every establishment shall maintain records in relation to employment, facilities provided and other necessary information with regard to compliance with the provisions of this Chapter in such form and in such manner as may be prescribed by the appropriate government;

- (2) These records shall specifically include information on women with disabilities in relation to their employment, facilities provided and other information as prescribed;
- (3) Every employment exchange shall register in accordance with prescribed procedure and thereby maintain records of persons with disabilities seeking employment. These records shall specifically include data on women with disabilities;
- (4) Such records shall be relevant and authentic evidence of a person with disability seeking unemployment allowance under Section 64 (6) (d) of this Act;
- **(5)** Any person authorized by the State Disability Rights Authority may inspect the records during the working hours of the establishment.

60. Vocational Training, Rehabilitation and Self-Employment

- (1) The appropriate governments shall take all necessary measures with respect to formulation of schemes and programmes such as Abylimpics to facilitate and support employment of persons with disabilities, with special reference to self-employment and vocational training of persons with disabilities;
- (2) Without prejudice to the general obligation stated in sub-section (1), the appropriate governments shall establish in each district work centres where persons with disabilities in rural areas can be imparted necessary skills and provided work opportunities in different trades including rural trades²⁹;
- (3) Without prejudice to the general obligation stated in sub-section (1), the appropriate governments shall ensure imparting of skills through convergence in existing training centers and institutions³⁰ and establish centers where none exists so that persons with disabilities in rural areas can be imparted necessary skills in crafts, trades and domiciliary occupations and provide work;
- (4) The appropriate governments shall provide adequate loans at concessional rates under the existing microcredit and loan schemes to persons with disabilities in order to facilitate self-employment schemes;
- (5) If, in the opinion of the person recruited, there is a need to impart specific training prior to recruitment in order to ensure that a person with disability has adequate support, then such facilities should be made available;

²⁹ As suggested at the Uttarakhand State Consultation

³⁰ As suggested at the Uttarakhand State Consultation

(6) The appropriate governments shall institute suitable schemes to promote and support the creative skills of persons with disabilities in rural and urban areas³¹ by establishing networks between the artisans and marketing federations and handicraft boards.

61. Incentives to Establishments

The appropriate governments shall within a period of one year of the enforcement of this Act provide incentives to all establishments to ensure that at least ten per cent of their work force is composed of persons with disabilities.

62. Grievance Redressal

- (1) Each establishment shall, as part of their Employment policy, set up a grievance redressal cell. The officer-in-charge of such grievance redressal cell shall be termed the grievance redressal officer, and shall be registered with the State Disability Rights Tribunal;
- (2) Any person may register a complaint with the grievance redressal officer, who shall maintain a register of complaints in the manner prescribed by the appropriate governments, and any such complaint shall be looked into within two weeks of such registration;
- (3) Without prejudice to the general nature of the above obligation, the grievance redressal officer shall ensure that every establishment shall fulfill the obligations stated in sub-section (4) of Section 56 In case of any lapse on the part of the establishment in the fulfillment of this obligation, the grievance redressal officer shall be liable unless he or she is able to demonstrate that the establishment has defaulted despite his or her efforts to ensure otherwise;
- **(4)** If the grievance redressal officer fails to address the complaint within two weeks of being registered, then the person with disability may approach the District Disability Rights Tribunal.

63. Special Employment Exchanges

³¹As suggested at the Maharashtra State Consultation

- (1) The appropriate governments may, retain the special employment exchanges established under Section 34 of the Persons with Disabilities (Equal Opportunities, Protection of Rights and Full Participation) Act, 1995 and establish new exchanges;
- (2) After the enforcement of this Act any notification issued by the appropriate government under Section 34(1) of the Persons with Disabilities (Equal Opportunities, Protection of Rights and Full Participation) Act, 1995 shall continue to be effective and establishments shall adhere to any new notifications issued by the appropriate government in this regard.

64. Right to Social Security

- (1) In furtherance of the right to life and living elaborated in Section 22 and the right to live independently and in the community in Section 24 of this Act, all persons with disabilities shall have a right to social security which includes but is not limited to securing adequate standard of living for persons with disabilities and their families in terms of food security, shelter, housing, social care, pension, unemployment allowance, health care, medical support, medical and life insurance;
- (2) The appropriate governments shall promulgate necessary schemes and programmes to safeguard and promote the realization of the rights of persons with disabilities recognized in sub-section (1) to provide adequate standard of living and improve their living conditions to enable them to live independently and in the community. In devising these schemes and programmes the diversity of disability, gender, age, and socio-economic status shall be relevant considerations;
- (3) These plans shall inter alia provide in rural, urban, tribal and hilly areas for:
 - a. safe and hygienic community centres with decent living conditions in terms of nutritional food, sanitation, health care and counseling;
 - b. facilities for persons including children with disabilities who have no families or have been abandoned, or are without shelter or livelihood;
 - c. support required during times of natural or man-made disasters and in areas of conflict:

- d. social care to women with disabilities who have no livelihood; and family support for upbringing of their children;
- e. a person with disability to be considered as a single unit for the purpose of availing benefit under any food security scheme;
- f. access to safe drinking water and appropriate and accessible 32 sanitation facilities especially in urban slums and rural areas.
- (4) The appropriate government shall formulate schemes to provide for social security benefits, aids and appliances, medicine and diagnostic, corrective surgery³³ without cost to persons with disabilities belonging to economically weaker sections of society;
- (5) Any refusal of insurance on the grounds of disability is unlawful. Any rule in standard form contracts that persons with disabilities constitute higher or unacceptable risk constitutes discrimination and is invalidated; and any practice to that effect stands invalidated:
- (6) In furtherance of right to live independently and in the community, the appropriate government shall inter alia:
 - a. frame disability pension schemes to cover persons with disabilities requiring support in all such schemes, the person with disability shall be the recipient for award of disability pension;
 - b. frame rules for making provisions for transfer of pension benefits of government employees to their children with disabilities;
 - c. provide for special initiatives such as enhanced disability pension for persons with multiple disabilities, persons needing high support, women and elderly persons with disabilities;
 - d. provide an unemployment allowance to all persons with disabilities proportionate to a uniform applied standard of qualification and skill; irrespective of the tenure of registration with the employment exchanges;
 - e. provide adequate care-giver allowance to all persons with disabilities meeting high support;
 - pay enhanced post retirement benefits to persons with disabilities;

³² As suggested at the Karnataka State Consultation ³³ As suggested at the Karnataka State Consultation

- g. organise special camps at Gram Panchayat level, at least once a year, to give information on, and to introduce persons with disabilities to, various government schemes.
- (7) The appropriate government shall, in furtherance of the right to habilitation and rehabilitation and the right to living independently, provide *inter alia:*
 - a. for not less than six percent reservation of land and housing to persons with disabilities in all government housing schemes in urban and rural development programmes with priority to women and elderly persons with disabilities;
 - b. ensure that not less than six percent of the beneficiaries, in all poverty alleviation and various developmental schemes, shall be persons with disabilities with priority to women and elderly persons with disabilities;
 - c. provide for not less than six percent reservation in allotment of land on concessional rate, where such land shall be used for the purpose of promoting housing, shelter, setting of occupations, businesses, enterprises, self help groups, gainful occupations, recreation centers, production centers exclusively for persons with disabilities whether or not such facilities are owned and managed by persons with disabilities;
 - d. to provide incentives to house owners by way of reduction in property tax who lease out premises to persons with disabilities for residential or commercial purposes;
- (8) No person with disability who is willing to work under the employment guarantee schemes or such other developmental schemes shall be denied work.

65. Right to Health

- (1) All appropriate governments and establishments shall ensure that:
 - a. all persons with disabilities enjoy the highest attainable standard of health on an equal basis with others and that persons with disabilities are not directly or indirectly discriminated on the grounds of disability;
 - b. all persons with disabilities are provided health care in accessible environments through accessible procedures and with reasonable

- accommodation. Domiciliary services shall be available wherever required to attain universal coverage;
- c. all schemes and programmes devised to realize health care rights, entitlements and benefits are universally available with due consideration accorded to gender, age and socio-economic status;
- d. persons with disabilities have access to free or affordable quality health care close to their communities particularly in the rural areas;
- e. health care services may be provided through multi-disciplinary teams who may undertake early identification, intervention and referrals to more specialized agencies wherever required.
- (2) In fulfillment of its obligation under this section the appropriate governments shall make schemes and programmes with the participation and involvement of persons with disabilities and care-givers that *inter alia* makes provision for:
 - a. the minimization and prevention of further disabilities with requisite education, training, information and intervention;
 - b. the health care of persons with disabilities during times of natural disasters and other situations of risk;
 - c. disability specific equipments and accessible infrastructure at all health care centres; public buildings and places; and all other such places that may be notified by the appropriate government from time to time;
 - d. 'essential medical facilities' for all life saving emergency treatment and procedures;
 - e. sexual and reproductive health especially of women with disabilities;
 - f. pre-natal, peri-natal and post natal care of mother and child;
 - g. nutritional intervention for children with disabilities³⁴;
 - h. psychosocial care and support at every stage of the medical process be it investigation, evaluation, diagnosis, treatment or intervention;
 - i. access to all Primary Health Care services especially within rural areas and for the urban poor³⁵;
 - j. free or subsidized treatment and medical services for weaker and indigent sections, as required;

³⁴ As suggested at the Karnataka State Consultation

³⁵ As suggested at the Maharashtra State Consultation

k. coverage of medical expenses, travel allowances and therapeutic intervention within a comprehensive insurance scheme for persons with disabilities.

Explanation

'Medical Services' include surgery, therapy, medicines, pathology and follow up.

66. Insurance

- (1) All establishments shall provide medical and life insurance to persons with disabilities on an equal basis with others;
- **(2)** The denial of such insurance or its provision on disproportionate premiums or iniquitous conditions would constitute discrimination.

67. Ethical Guidelines

- (1) All health professionals shall not discriminate against persons with disabilities in the observance of ethical guidelines on informed consent and confidentiality whilst providing health care and other services to persons with disabilities;
- (2) Without prejudice to the general obligation stated in sub-section (1) all health care professionals shall when making impairment specific interventions take special care to provide complete information to persons with disabilities through accessible modes, methods and formats:
- (3) Any negligent departure from the ethical guidelines shall invite action for professional misconduct or poor and inadequate service;
- **(4)** The medical practice guidelines shall be suitably revised to incorporate accessible procedures of communicating with persons with disabilities and delete any prejudicial references.

68. Awareness Raising Responsibilities

- (1) The appropriate governments and concerned establishments in collaboration with the National Disability Rights Authority and the State Disability Rights Authority, as the case may be, shall carry out sensitization and awareness-raising amongst care-givers, service-users and the general public as regards the human rights, dignity, autonomy and needs of persons with disabilities;
- (2) The appropriate governments and concerned establishments shall:

- a. Undertake or cause to be undertaken surveys, investigation and research on the health status and needs of persons with disabilities;
- b. Provide for training of primary health care centres personnel and paramedics³⁶ to address the needs of persons with disabilities;
- c. Sponsor or cause to be sponsored awareness campaigns and disseminate or cause to be disseminated information on general hygiene, health and sanitation as impacting the health status of persons with disabilities.

69. Prohibition on Denial of Food and Fluids

- (1) The denial of foods and fluids to persons with disabilities including infants with disabilities by reason of their disability is hereby prohibited;
- (2) Any person found to either deny food and fluids to any person with disability or to aid and abet in such denial shall be liable to criminal prosecution and penalties under Section 155. Any contravention of this Section shall be penalized under Section 155.

70. Right to Habilitation

(1) In furtherance of the right to life recognized by Article 21 of the Constitution of India and Section 22 of this Act all persons with disabilities have a right to habilitation in order to realize the rights guaranteed under this Act;

Explanation

Habilitation is a process by which persons born with impairments learn life skills. Rehabilitation refers to a process by which persons readapt to society after acquiring impairment.

(2) The appropriate governments and establishments shall in furtherance of the right recognized in sub-section (1) take effective and appropriate measures to enable persons with disabilities to maximize their physical, mental, personal, social, educational, vocational and professional abilities to enable them to live a full, inclusive and dignified life. In devising these measures the diversity of disability; gender; age; and socio-economic status shall be relevant considerations;

³⁶As suggested at the Uttarakhand State Consultation

- (3) Without prejudice to the generality of the obligation in sub-section (2), in order to ensure that children with disabilities have the same life development opportunities as other children, the appropriate governments and local authorities shall formulate habilitation plans for children with disabilities with strategies for ensuring that it starts as early as possible;
- (4) These plans shall inter alia provide for:
 - a. Appropriate support and guidance to parents of infants and young *c*hildren with disabilities to enable them to perform their parenting responsibilities;
 - b. The training of children and young persons with disabilities in self care and self reliance depending upon the nature of impairment;
 - c. The promotion of total communication among children and adults with different types of impairments;
 - d. Provision of social skills and social networking between children with disabilities; other children; extended family and larger community;
 - e. Therapeutic, surgical or other corrective interventions and for aids and appliance;
 - f. Ensuring the provision of adequate training in mobility, orientation of environment, use of assistive devices and application of technology.
- (5) In order to ensure that every child with disability obtains access to habilitation at the earliest possible time the appropriate government shall:
 - a. Proactively survey the presence of infants and children with disabilities;
 - disseminate information on nature of disabilities and the plan of habilitation to the community generally; and to community workers, doctors and teachers particularly.
- **(6)** The habilitation plan shall have a dynamic approach and would change in its interventions and outcomes in accordance with the age, gender, local environment and functional goals of the persons with disabilities;
- (7) Every person with disability shall have the right to take a decision with regard to their own habilitation plan; and due consideration should be given to the evolving capacity of children in supporting and obtaining their participation.

71. Right to Rehabilitation

- (1) All persons with disabilities have the right to live a full and meaningful life irrespective of the nature of the impairment and the manner, gender, age, caste,³⁷ place, situation or circumstance in which such impairment is acquired;
- (2) In furtherance of this right to life and living all persons with disabilities have a right to peer, professional, community and state support to resume and continue their personal, social, economic and political participation on an equal basis with others;
- (3) In enforcement of this abovementioned right the appropriate governments and local authorities shall put in place suitable peer and professional intervention to enable the persons with disabilities to accept the altered nature of their mind and body;
- (4) All persons with disabilities have a right to be provided aids and appliances of recognized quality at an affordable cost along with the requisite training to utilise it:
- **(5)** Every person with disability has the right to be informed of the various rehabilitation options and make the final decision on the course of rehabilitation;
- **(6)** The appropriate governments and concerned establishments shall put in place suitable infrastructure and expertise for the rehabilitation of persons with disabilities.

72. Research and Development

The appropriate governments and the National and State Disability Rights Authority as the case may be shall initiate or cause to initiate research and development through individuals and institutions on issues which would enhance the habilitation and rehabilitation processes for persons with disabilities.

73. Right to Leisure, Culture and Recreation

(1) All appropriate governments and establishments shall take all suitable measures to promote and protect the right of all persons with disabilities to have a cultural life and to participate in leisure and recreational activities on an equal basis with others;

³⁷ As suggested at the Maharashtra State Consultation

- (2) Without prejudice to the general obligation in sub-section (1) such measures shall include
 - a. Providing facilities support and sponsorships to artists and writers with disabilities to pursue their interests and talents;
 - b. Sponsoring of disability film, theatre, music and dance festivals;
 - c. Establishment of a disability history museum which recovers, chronicles and interprets the historical experiences of persons with disabilities;
 - d. Making art accessible to persons with disabilities;
 - e. Promoting the launch of leisure clubs, recreation centers, and other associational activities;
 - f. Facilitating participation in scouting, dancing, art classes, and outdoor camps;
 - g. Redesigning courses in cultural and arts subjects to enable participation and access of persons with disabilities;
 - h. Developing technology, assistive devices and equipments to facilitate access and inclusion for persons with disabilities in recreational activities.
- (3) All Cultural Academies whether of, art, literature, music, or dance shall include persons with disabilities in their programs schemes, and activities. And shall thereby provide recognition, support, and awards to the cultural contribution of persons with disabilities on an equal basis with others.

74. Right to Participation in Sports, Games and Athletics

- (1) All persons with disabilities have a right to participate in sports at all levels on an equal basis with others;
- (2) All appropriate governments shall take appropriate measures to guarantee the effective participation in sporting activities for all persons with disabilities. Such measures shall expressly provide for the inclusion of persons with disabilities in all mainstream sporting events and activities;
- (3) All National and State Sports bodies shall accord due recognition to the right of all persons with disabilities to participate in sports and shall make due provision for the inclusion of persons with disabilities in all their schemes and programs for the promotion and development of sporting talent;

- (4) The Central Government shall establish a National Body for Disability Sports to promote the sports of persons with disabilities³⁸;
- (5) All international and national sporting events for persons with disabilities including Special Olympics and Paralympics shall be accorded the status due to international and national sporting events and the sportspersons participating in such events shall be allocated resources, sponsorships, awards and jobs on an equal basis with others³⁹;
- **(6)** Without prejudice to the general obligations stated in sub-section (2) and (3), the appropriate governments and the National and State Sports bodies shall take specific measures including to:
 - a. restructure courses and programmes to ensure access, inclusion and participation of persons with disabilities in all sporting activities;
 - b. redesign and support infrastructure facilities of all sporting activities for persons with disabilities;
 - c. develop technology to enhance potential, talent, capacity and ability in sporting activities of all persons with disabilities;
 - d. provide multi-sensory essentials and features in all sporting activities to ensure effective participation of all persons with disabilities;
 - e. allocate funds for development of state of art sport facilities for training of persons with disabilities;
 - f. promote and organize disability specific sporting events for persons with disabilities.

PART II: POWERS, DUTIES AND RESPONSIBILITIES

75. Awareness Raising

(1) All appropriate governments in consultation with the National or State Disability Rights Authority as the case may be shall conduct, sponsor, encourage, support

³⁸ As suggested at the Manipur State Consultation

³⁹ As suggested at the Orissa State Consultation

or promote on a regular and continuous basis⁴⁰ information campaigns and sensitization programmes to ensure that the rights recognized in this legislation are respected, protected and promoted. Such campaigns should aim at enabling both state and civil society to comprehend disability as an integral part of the human condition; to recognize the capabilities and contributions of persons with disabilities; and to combat the stereotypes, prejudices and harmful practices which impede the participation of persons with disabilities on an equal basis with others;

- (2) Without prejudice to the general objectives of the information campaigns and sensitization programmes stated in sub section (1), the appropriate governments shall ensure that the aforementioned awareness programmes especially draw attention to the direct and indirect discrimination faced by women and children with disabilities;
- (3) Subsequent to the coming into force of this Act the nomenclature and terminology developed in this legislation shall be employed in all rules, regulations, notifications, orders and in all communications and transactions by and with the State;
- (4) The appropriate governments shall launch and support stigma reduction programmes; mass education campaigns; information and technology linked dissemination programmes; and sensitization workshops on disability rights to achieve the objectives outlined in subsection (1) which amongst others shall address the educational authorities⁴¹; medical fraternity; the family; government officials; policing authorities; armed forces⁴²; political representatives; the media and the legal community;
- (5) In order to achieve the objectives outlined in subsection (1) every establishment shall undertake suitable sensitization and information dissemination programmes on disability rights within their organization; and may launch and support stigma reduction programmes, mass education campaigns, information and technology linked dissemination programmes;
- **(6)** Without prejudice to the general awareness raising obligation in sub section 2 such programmes, campaigns and workshops shall *inter alia:*
 - a. Promote values of inclusion, tolerance, empathy and respect for diversity;

⁴¹As suggested at the Delhi-Haryana State Consultation

⁴⁰As suggested at the J&K State Consultation

⁴²As suggested at the Manipur State Consultation

- Advance recognition of the skills, merits and abilities of persons with disabilities and of their contributions to the workforce, labour market and professional fields;
- c. Foster respect for the decisions made by persons with disabilities on all matters related to family life, relationships, bearing and raising children;
- d. Provide orientation and sensitization at the school, college, university and professional training level on the human condition of disability and the rights of persons with disabilities;
- e. Provide orientation and sensitization on disabling conditions and Rights of Persons with Disabilities to employers, administrators and co-workers⁴³.
- (7) The appropriate governments and establishments shall launch legal literacy and information dissemination programmes on disability rights generally and this law more particularly.

76. Accessibility

- (1) All persons with disabilities have the right on an equal basis with others to the physical environment, transportation, information and communications, including appropriate technologies and systems, and other facilities and services open or provided to the public, both in urban and in rural areas;
- (2) All appropriate governments and establishments shall be required to ensure that the rights of persons with disabilities to accessibility as outlined in sub section (1) of this section are fully protected.

77. Accessibility Standards

(1) The National Disability Rights Authority shall formulate regulations which lay down the standards of accessibility for the physical environment, transportation, information and communications, including appropriate technologies and systems, and other facilities and services open or provided to the public, both in urban and in rural areas;

 $^{^{\}rm 43}$ As suggested at the Karnataka State Consultations.

- (2) The National Disability Rights Authority shall formulate these Regulations, in consultation with subject and experiential experts, by adopting or adapting prevailing international standards on physical environment and information and communication technologies and systems to local conditions. It shall ensure that the regulations are age and gender appropriate and are applicable to:
 - a. All buildings and facilities used by the public;
 - b. Permanent, temporary or emergency conditions;
 - c. All new community residential places and private residences to make them visitable and live-able;
 - d. Pedestrian infrastructure;
 - e. Ports;
 - f. Road based transport; Aviation; Railways; Maritime transport; Rural Public Transport System; and all other modes of transport so as to ensure that persons with disabilities travel in safety and comfort.
- (3) The National Disability Rights Authority shall within a period of one year develop and notify the aforesaid accessibility standards regulations for both urban and rural areas. It shall review these regulations every five years and revise them if required in order to ensure universal coverage of all transport and built environment;
- (4) The appropriate governments and establishments shall take suitable measures:
 - a. to provide facilities for persons with disabilities at stations and airports that meet the accessibility standards relating inter alia to parking spaces, toilets, ticketing counters and ticketing machines;
 - b. to provide access to all modes of transport that conform to design standards, including retrofitting old modes of transport;
 - c. to formulate rules for providing driving licenses to persons with disabilities;
 - d. to ensure that where no modifications are possible, human assistance shall be made available to persons with disabilities;
 - e. to provide support services to assist persons with disabilities to negotiate and interact with unfamiliar people, environs and spaces;
 - f. to provide for rural modes of transport and accessible roads⁴⁴ to address mobility needs of persons with disabilities in rural areas.

⁴⁴As suggested at the Karnataka State Consultation.

78. Personal Mobility

- (1) All appropriate governments and establishments shall develop schemes and programmes to promote the personal mobility of persons with disabilities at affordable cost, according to their choice;
- (2) Such schemes may inter alia provide:
 - a. incentives and concessions to provide accessible transport facilities to persons with disabilities;
 - b. for retrofitting of vehicles for persons with disabilities at no extra cost;
 - c. appropriate personal mobility assistance;
 - d. rural modes of transport to address mobility needs of persons with disabilities in rural areas.

79. Access to services

- (1) All appropriate governments and establishments shall ensure that all services and facilities provided by them are available to persons with disabilities on an equal basis with others; and that such services are provided in such mode or format which is responsive to the needs of persons with disabilities but at no extra cost to them;
- (2) All appropriate governments and establishments shall ensure that all announcements inviting public response including those made for procurement, entitlement, employment, public health and disaster preparedness must be accessible to all persons with disabilities;
- (3) The National and State Disability Rights Authority shall make available to all service providers information on how they can make their services accessible to persons with disabilities.

80. Access to Information and Communication Technology

- (1) All appropriate governments and establishments shall take measures to ensure that:
 - a. All content in whichever medium whether audio, print or electronic shall be made available to persons with disabilities in accessible format;
 - b. Persons with disabilities shall have access to electronic media by providing for audio description, sign language interpretation and close captioning;
 - c. Accessibility to telecommunication services where telecommunications will include any kind of transmission of information of the user's choosing without change in form or content of information as sent or received;⁴⁵
 - d. Electronic goods and equipment of everyday use shall follow the principles of universal design;
 - e. Schemes are formulated or amended to ensure affordable access to Information and Communication Technology & Electronics for persons with disabilities in rural as well as urban areas;
 - f. All government websites, and private websites providing consumer services conform to the most updated version of the World Wide Web Consortium web

⁴⁵As suggested at the Legal Consultation

- accessibility standards;
- g. Incentives and concessions are provided to support existing websites to make them accessible to persons with disabilities.
- **(2)** The conversion, reproduction, adaptation and communication of all copyrighted materials into accessible versions for exclusive use of persons with disabilities on 'not-for-profit⁴⁶' basis shall be deemed fair dealing;

Explanation

Accessible version includes any version or form which gives a person with disability access to the work as flexibly and comfortably as a person without a disability, and shall include, but not be limited to, audio recordings, audio –visual works with audio or text descriptions, Braille, e-text including DAISY formats, digital copies compatible with assistive technology or refreshable Braille, tactile material, large print, with different type-faces and sizes and sign language.

- (3) All government and private websites shall be made accessible, in accordance to the regulations formulated by the National Disability Rights Authority, within a maximum period of one year from the date of notification of such regulations;
- (4) Any contravention of this Section shall be penalized under Section 150⁴⁷

81. Access to Consumer Goods and Services

- (1) All appropriate governments and establishments shall:
 - a. take measures to promote development, production and distribution of universally designed consumer products and accessories for general use;
 - b. take measures to provide accessible consumer services including personal grooming, fitness and beautification services.

82. Service Animals

(1) The National Disability Rights Authority shall formulate regulations for Service Animal training facilities so as to ensure that persons with disabilities are provided suitable Service Animals⁴⁸;

.

⁴⁶ As suggested at the Legal Consultation

⁴⁷ As suggested at the Legal Consultation.

- (2) The appropriate governments and establishments shall permit and facilitate the use of Service Animals by persons with disabilities on roads, buildings, all transport systems, public facility or service;
- (3) A person with disability needing assistance shall have a right to be accompanied by a Service Animal without being required to pay an extra charge for the Service Animal.

83. Mandatory Observances of Accessibility Norms

- (1) No individual, organization or establishment shall be granted permission to build any structure if the building plan does not adhere to the regulations formulated by the National Disability Rights Authority;
- (2) No individual, organization or establishment shall be issued a certificate of completion or allowed to take occupation of a building if it has failed to adhere to the regulations formulated by the National Disability Rights Authority;
- (3) Any contravention of this Section shall be penalized under Section 156.

84. Time Limit for Making Existing Infrastructure and Premises Accessible

- (1) All existing public buildings used for government purposes shall be made accessible in accordance with the regulations formulated by the National Disability Rights Authority, within a period not exceeding three years from the date of notification of such regulations;
- (2) All other public buildings shall be made accessible in accordance with regulations formulated by the National Disability Rights Authority, within a period not exceeding five years from the date of notification of such regulations;
- (3) Any contravention of this Section shall be penalized under Section 150.

85. Time Limit for Accessibility Duties of Service Provider

(1) All service providers shall provide services in accordance with the regulations on accessibility formulated by the National Disability Rights Authority within a period of one year from the date of notifications of such regulations;

⁴⁸As suggested at the Karnataka State Consultation.

(2) Any contravention of this Section shall be penalized under Section 157.

86. Role of the National Centre for Universal Design and Barrier-free Environment

- (1) The Central Government shall establish the National Centre for Universal Design and Barrier Free Environment in order to assist the country to become universally accessible and inclusive in terms of accessibility;
- (2) The National Centre shall support the National and State Disability Rights Authority in developing and monitoring Accessibility regulations for all services and facilities provided to the public at large both in the rural and urban areas;
- (3) In furtherance of the role outlined in sub-section (1) the National Centre shall undertake the following functions:
 - a. Research and Development in the field of universal design and barrier free environment:
 - Develop curriculum in the field of architecture, design, technology, electronics and engineering in order to incorporate the theory and practice of universal design;
 - c. Facilitate students in educational institutions to undertake projects pertaining to universal design;
 - d. Provide consultancy services to all establishments and individuals on such request.

Provided that the National Centre shall be the permanent consultant on all issues of universal design and barrier free environment to the National and State Disability Rights Authorities and the appropriate governments.

87. Implementation Plan

- (1) The National and State Disability Rights Authorities in collaboration with NIUD shall formulate an action plan based on prioritization of providing accessibility in all public buildings and spaces providing essential services such as all Primary Health Centres, Civil/District hospitals, Primary schools/ secondary schools, railway stations and bus stations within three years;
- (2) A plan shall include carrying out of access audits by authorized team of auditors and provision of accessible facilities based on standard accessibility guidelines;
- (3) All plans shall be public documents and shall be made available by the National

- and State Disability Rights Authorities in formats accessible to persons with disabilities:
- (4) The National and State Disability Rights Authorities shall set up a system of monitoring all existing built infrastructure once the plans to provide accessibility have been made public.

88. Human Resource Development

- (1) The appropriate governments shall ensure that the social, economic and civil political rights guaranteed in this Act are duly implemented and to that end shall:
 - a. undertake the development of human resource so that both inclusive and specialized services are made available to persons with disabilities⁴⁹;
 - b. orient, sensitize, and train⁵⁰ existing personnel and create curricula which are disability sensitive;
 - c. make provision for adequate numbers of professionals in such manner that such personnel are available in appropriate ratios to provide services for persons across all disabilities at the central, state, local and panchayat level;
 - d. the provision of personnel to monitor the conduct of training at central, state, district and local bodies and panchayat levels to ensure effective functioning of institutions and professionals.
- (2) In order to fulfill the obligation stated in sub section(1) the appropriate governments shall every three years⁵¹ undertake a needs based analysis and formulate plans for the recruitment, induction, sensitization, orientation and training of suitable personnel to undertake the various responsibilities outlined in this Act;
- (3) The appropriate governments shall ensure that the terms and conditions of service of all personnel, professionals, workers and any other person employed in whatsoever manner and in any capacity⁵² in the field of disability are

⁴⁹ As suggested at the Uttaranchal State Consultation

⁵⁰ As suggested at the Karnataka State Consultation

⁵¹As suggested at the Karnataka state consultation.

⁵² As suggested at the Uttaranchal State Consultation

- appropriate, just and equitable so as to ensure the induction and retention of quality personnel and professionals in the field in adequate numbers;
- (4) Without prejudice to any function and power of Rehabilitation Council of India contained in the Rehabilitation Council of India Act and in addition to the cadre creating obligations outlined in sub section (2) the appropriate governments and establishments⁵³ shall ensure human resource development in this sector by *inter alia:*
 - a. Mandating training on disability rights in all courses for the training of panchayati raj members⁵⁴. legislators⁵⁵, administrators, police officials, judges, lawyers and judicial officers:
 - Induction of disability as a component for all education courses/ programmes for school, college and university teachers, doctors, nurses, para-medical personnel, social welfare officers, rural development officers, Aasha workers, Anganwadi workers engineers, architects, other professionals and community workers;
 - c. Initiating capacity building programmes, including training in independent living and community relationships for families, members of community and other stakeholders and care providers on care giving and support;
 - d. Ensuring independence training for persons with disabilities to build community relationships on mutual contribution and respect;
 - e. Conducting training programmes for sports teachers with specific focus on sports, games, athletics and skill demonstration of persons with disabilities.
- (5) All Universities shall promote teaching and research in disability studies including establishment of dedicated centres for such studies⁵⁶.

89. Financial Assistance to Registered Organizations

All appropriate governments shall grant financial assistance to registered organizations to provide services and to implement the schemes and programmes mandated under this Act.

⁵³ As suggested at the West Bengal State Consultation

⁵⁴ As suggested at the Odisha State Consultation

⁵⁵ As suggested at the Karnataka State Consultation

⁵⁶ As suggested at the Karnataka consultation

90. Disability Cell

There shall be set up a Disability Rights Cell in all appropriate governments and establishments to coordinate the implementation of this Act.

91. Disability Audit

All appropriate governments shall be under a duty to undertake an audit of all general schemes and programs in order to ensure that they do not have an adverse impact upon persons with disabilities and to induct the requirements and concerns of persons with disabilities⁵⁷.

92. Power to Amend Schedule

- (1) The Central Government may keeping in view the evolving nature of disability periodically constitute a committee with equal representation of subject and experiential experts to determine whether any additions need to be made in the list of impairments included in Schedule 1;
- (2) The Committee constituted under sub section (1) shall develop a socio-medical scale to examine the interplay between a particular impairment and the barriers. The severity of the impairment and the scale of the barrier experienced by persons with disabilities shall guide the Committee in making its recommendations.
- (3) The central government may suitably incorporate the socio-medical scale in the normative descriptions provided in Schedule 1.

93. Disability Certificates

- (1) The appropriate governments shall designate in every panchayat, municipal and notified area, officials or authorities who would be authorized to issue a disability certificate to persons with disabilities in accordance with prescribed procedure;
- (2) The prescribed procedure shall be devised so as to facilitate the obtaining of the certificate by persons with disabilities. The duration of the certificate shall be

⁵⁷ As suggested at the Legal Consultation

- determined having regard to the nature of the impairment and a certificate issued by any one authority shall be given full faith and credit throughout the territory of India⁵⁸:
- (3) Notwithstanding any modification of standards in Schedule 1 the disability certificates obtained by persons with disabilities under the Persons with Disabilities (Equal Opportunities Protection of Rights and Full Participation) Act of 1995 (Act No 1 of 1996) or under this Act prior to such modification shall continue to remain valid for the period for which they have been issued.

PART III: REGULATORY AND MONITORING AUTHORITIES

94. Establishment and Incorporation of National Disability Rights Authority

- (1) There shall be established, for the purposes of promoting, protecting and ensuring the full and equal enjoyment of all human rights and fundamental freedoms of all persons with disabilities, an Authority which shall be called the National Disability Rights Authority;
- (2) National Disability Rights Authority shall be a body corporate by the name aforesaid, having perpetual succession and a common seal, with power, subject to the provisions of this Act, to acquire, hold and dispose of property, both movable and immovable, and to contract, and shall, by the said name, sue or be sued;
- (3) National Disability Rights Authority shall have its head and seat in New Delhi.

95. Composition and Management of the Authority

(1) National Disability Rights Authority shall consist of a Chairperson and a Governing Board comprising of:

⁵⁸ As suggested at the Legal Consultation

a. 12 members of whom at least 7 shall be full-time members who are persons with disabilities and shall be appointed by way of an election, by an electoral college that comprises of Disabled Persons Organizations, the organizations of parents and families of persons with disabilities and organizations working for the rights of persons with disabilities which are registered under this Act or the National Trust Act, 1999, by such procedure as may be prescribed by the Central Government;

Provided that the members of the first Governing Board shall be nominated by the President on obtaining recommendations of the Selection Committee under Section 100 from amongst the panel of names forwarded by Disabled Persons Organizations, the organizations of parents and families of persons with disabilities and organizations working for the rights of persons with disabilities.

- b. 5 members not below the rank of Joint-Secretary to the Government of India and shall be nominated by the Central Government, representing the Ministry of Social Justice and Empowerment (Disability Division), Health and Family Welfare, Urban Development, Rural Development and Human Resource Development;
- c. 2 members who are from National Institutes on annual rotation represented by the Director of the National Institute;
- d. 2 members nominated by the Federation of Indian Chambers of Commerce and Industry the Confederation of Indian Industries; and
- e. a Member-Secretary with requisite knowledge and experience in the field of disability to be appointed in accordance with the procedure prescribed.
- (2) The Chairperson may invite a representative of any other Ministry or Department or any professional or technical expert, when deemed necessary, for due deliberation on any matter, before the National Disability Rights Authority;
- (3) The Chairperson of the National Trust and the Chairperson of the Rehabilitation Council of India shall be ex-officio members of the Board;
- (4) The Governing Board may, by simple majority, appoint one of the members to be Vice-Chairperson who shall exercise and discharge such powers and functions of the Chairperson as may be prescribed and as may be delegated by the Chairperson;
- **(5)** The Chairperson shall be a full time member;

- (6) The Chairperson and the fulltime Members of the National Disability Rights Authority shall not hold any other office of profit;
- (7) Any person elected as member to both the National Disability Rights Authority and State Disability Rights Authority shall resign from the membership of any one of them in accordance to his or her choice.

96. Powers, Functions and Meetings of Governing Board⁵⁹

- (1) The Governing Board shall have the plenary powers of the National Disability Rights Authority. It shall envision the broad policies, and programmes of the Authority and periodically review them; and devise measures for the effective and efficient functioning of the Authority and shall also;
 - a. consider and pass the Plan of Action, regulations of practice prepared by the Executive Board and to adopt them with or without modification;
 - b. sign Memorandum of Understanding, forging national and international partnership and co-operation;
 - c. consider and pass the annual report, financial statement and the budget estimates prepared by the Executive Board and to adopt them with or without modification.
- (2) The Governing Board shall meet at least once in a year and shall hold an annual general meeting with Electoral College at the head office or any other place at such time as the Chairperson may direct, and shall observe such rules of procedure in regard to the transaction of business at its meetings (including the quorum at its meetings) as may be specified;
- (3) When the Chairperson is unable to discharge the functions of office owing to absence, illness or any other cause, the Vice-Chairperson shall discharge the functions of the Chairperson until such date that the Chairperson resumes charge;
- (4) All questions which come up before any meeting shall be decided by a majority of votes of the Members present and voting, and in the event of an equality of votes, the Chairperson or the person presiding shall have the right to exercise a second or casting vote;

⁵⁹ As suggested at the Legal Consultation

- (5) All orders and decisions of the Governing Board shall be authenticated by the Member-Secretary or any other officer duly authorised by the Chairperson in this behalf:
- (6) No act or proceedings of the Authority shall be questioned or invalidated merely on the ground of existence of any vacancy or any defect in, the constitution of, the National Disability Rights Authority.

97. Executive Board

- (1) The National Disability Rights Authority shall operate through an Executive Board which shall be constituted from amongst the members of the Governing Board by such procedure as may be prescribed;
- (2) The Executive Board shall consist of the Chairperson and nine other members. The Chairperson of the Governing Board shall be the Chairperson ex-officio of the Executive Board; of the nine other members: five members from amongst the elected members; two members from Government; one member from the National Institutes and one member from commerce and industry; on deputation by such procedure as may be prescribed;
- (3) The Member-Secretary of the Governing Board shall be the Member-Secretary of the Executive Board, ex-officio;
- (4) The Executive Board may, by simple majority, appoint one of the members to be Vice-Chairperson who shall exercise and discharge such powers and functions of the Chairperson as may be prescribed and as may be delegated by the Chairperson.

98. Powers, Functions and Meetings of the Executive Board

- (1) The general superintendence, direction and management of the affairs of National Disability Rights Authority shall vest in the Executive Board of members, which may exercise all powers and do all acts and things which may be exercised or done by National Disability Rights Authority;
- (2) The Executive Board shall have such powers and functions:
 - a. to prepare and present a Plan of Action to the Governing Board;
 - b. to formulate and set standards by way of regulations;

- c. to ensure compliance of regulations and initiate inquiries in alleged violation or non-compliance of regulations;
- d. to prepare and present to the Governing Board at its annual meetings, report on the working of the Authority; a statement of accounts; and budget proposals for the awareness raising, capacity building and sensitization programmes;
- e. to promote research, collection of data and material;
- f. to select a common seal for the Authority; and
- g. to exercise such other powers and to perform such other duties as may be considered necessary; or imposed on it by or under this Act.
- (3) The Executive Board shall hold meetings at the head office or any other place at such time as the Chairperson may direct, and shall observe such rules of procedure in regard to the transaction of business at its meetings (including the quorum at its meetings) as may be specified;
- (4) When the Chairperson is unable to discharge the functions of the office owing to absence, illness or any other cause, the Vice-Chairperson shall discharge the functions of the Chairperson until such date that the Chairperson resumes charge;
- (5) All questions which come up before any meeting shall be decided by a majority of votes of the Members present and voting, and in the event of an equality of votes, the Chairperson or the person presiding shall have the right to exercise a second or casting vote;
- **(6)** No act or proceedings of the Authority shall be questioned or shall be invalidated merely on the ground of existence of any vacancy in, or any defect in, the constitution of the National Disability Rights Authority.

99. Reservation within the Elected Members of the Governing Board

- (1) Of the 12 members comprising of persons with disabilities on the Governing Board, the following reservations shall be provided:
 - a. 1 seat shall be reserved for persons with autism and intellectual disabilities;
 - b. 1 seat shall be reserved for persons with mental illness;

- c. 1 seat shall be reserved for persons with multiple disabilities, deaf-blindness, multiple sclerosis;
- d. 2 seats shall be reserved for women with disabilities.
- (2) The provisions of sub section (1) shall be applicable only in the event and to the extent that the constituencies mentioned therein, do not find representation by way of general election;
- (3) In the event that the persons with disabilities under clause (a) of sub-section (1) or clause (b) of sub-section (1) are not available to stand for election, the seats so reserved shall be open for election to the care-givers of persons with the respective disabilities.

100. Selection Committee for the Chairperson

- (1) The Chairperson of National Disability Rights Authority shall be appointed by warrant under the hand and seal of the President of India; provided that every appointment shall be made on obtaining the recommendations of Selection Committee, from amongst the panel of names forwarded by Disabled Persons Organizations, Organizations of Parents of persons with disabilities and their families and organisations working for the rights of persons with disabilities;
- (2) For the purpose of sub-section (1), the Selection Committee shall comprise of:
 - a. the Prime Minister of India;
 - b. the Leader of the Opposition in the Lok Sabha;
 - c. the Leader of the Opposition in the Rajya Sabha;
 - d. the Union Minister for Social Justice and Empowerment;
 - e. the Union Minister for Corporate Affairs;
 - f. two experts who shall be persons of integrity and standing who have special knowledge of and experience in the field of disability, law, health, education, management, industry, economics, business, commerce and public affairs.
- (3) No appointment of Chairperson or other Member shall be invalid merely by reason of any vacancy in the Selection Committee.

101. Term of Office of Chairperson and Members

- (1) A person appointed as Chairperson shall hold office for a term of five years from the date of entering office or until seventy years, whichever is earlier;
- (2) A person appointed as a Member shall hold office for a term of five years from the date of entering office and shall be eligible for re-appointment for another term of five years; no person shall be appointed as member of National Disability Rights Authority for more than two consecutive terms or hold office after attaining the age of seventy years.

102. Terms and Conditions of Service of Chairperson and Members

The salaries and allowances payable to, and other terms and conditions of service of, the Chairperson and Members shall be such as may be prescribed by the Central Government;

Provided that neither the salary and allowances nor the other terms and conditions of service of the Chairperson or a Member shall be varied to his or her disadvantage after his or her appointment

103. Officers and Other Staff of the National Disability Rights Authority

- (1) Subject to such rules as may be made by the Central Government the National Disability Rights Authority may appoint such other administrative, technical and scientific staff as it may consider necessary;
- (2) The salaries, allowances and conditions of service of the officers and other staff appointed under sub-section (1) shall be such as may be prescribed by the Central Government.

104. Resignation, Removal and Suspension of the Chairperson and Other Members

- (1) The Chairperson may resign office by tendering notice in writing to the President of India.
- (2) Any other member may resign office by tendering notice in writing to the Chairperson.

- (3) The President of India may remove the Chairperson from office if he or she:
 - a. has been adjudged a insolvent; or
 - b. has been convicted of an offence which involves moral turpitude; or
 - c. has so abused the position as to render continuance in office prejudicial to public interest.
- **(4)** The Governing Board may, by a resolution passed by two-third majority, remove from office any member, who:
 - a. has been adjudged a insolvent; or
 - b. has been convicted of an offence which involves moral turpitude; or
 - c. has abused the position to render continuance in office prejudicial to the public interest.
- (5) No such member shall be removed from office under clause (c) or clause (d) of sub-section (3) and under clause (c) or clause (d) of sub-section (4) unless given a reasonable opportunity of being heard in the matter;
- (6) In the event of a vacancy in the office of the Chairperson by reason of death, resignation, suspension or removal; the Vice-Chairperson shall act as the Chairperson, until the date on which a new Chairperson, appointed in accordance with the provisions of this Act to fill such vacancy, enters upon office.

105. Functions of the National Disability Rights Authority

- (1) Subject to the provisions of this Act, it shall be the duty of the National Disability Rights Authority to promote, protect and monitor the enforcement of the rights of persons with disabilities by such measures as it deems fit;
- (2) Without prejudice to the generality of the foregoing provisions, the National Disability Rights Authority, to fulfil the objective outlined in sub-section (1) may inter alia provide for
 - a. assistance in developing an effective and comprehensive policy and regulatory framework to address issues faced by persons with disabilities;
 - b. reviewing and evaluating legislations, policies, rules, regulations, bye-laws, programmes, notifications or any other legal instrument at the Union, State

- and local level on a periodic basis to determine their conformity with rights of persons with disabilities;
- c. calling from or furnishing to or exchanging information for the efficient discharge of its functions with such agencies, regulatory authorities or enforcement bodies as required;
- d. co-ordinating and reviewing activities of appropriate government to ensure proper coverage of persons with disabilities in their activities and programmes with adequate resource allocation;
- e. making recommendations to appropriate government with regard to specific budgetary allocations by all its departments in their respective budgets outlays every year⁶⁰;
- f. monitoring utilization of funds allocated for persons with disabilities⁶¹;
- g. preparing and publishing a Plan of Action within six months of the date of formation of National Disability Rights Authority that shall identify principles and priorities and shall make a copy of the said Plan of Action, available to State Disability Rights Authority that shall adapt the same to accommodate the state specific priorities and requirements;
- h. listing the activities to be undertaken by the National Disability Rights Authority for a *two-year period* with provision for review and revision as and when necessary;
- i. developing and conducting information programmes to foster public understanding and guidance on this Act; the Convention; and the role and activities of National Disability Rights Authority so as to sensitize and raise awareness and foster respect for the rights and dignity of persons with disabilities throughout society, including at the family level;
- j. promoting understanding, capacity building and the development and exchange of good practices by forging partnerships with public and private sector organisations, employers and service providers, NGOs and publicspirited individuals;
- k. undertaking research, collecting data and materials relating to persons with disabilities and disseminating the same to promote evidence based understanding of disability rights;

⁶¹ As suggested at the Madhya Pradesh State Consultations

^

⁶⁰ As suggested at the Uttarakhand State consultations

- I. issuing binding Regulations on the standards of practice by National Disability Rights Authority, a copy of which shall be made available to State Disability Rights Authority, who can adapt them, as required, for the state;
- m. ensuring effective compliance of regulations;
- n. co-ordinating and providing assistance and expert advice to State Disability Rights Authority in effective and efficient discharge of its functions including developing and conducting information programmes, sensitization and awareness raising with regards to the Convention;
- o. inquiring, suo motu or on receiving a complaint, into any non compliance of regulations, rules, orders notification under the Act; and
- p. performing any other functions that are incidental and ancillary thereto
- (3) National Disability Rights Authority shall undertake functions outlined in subsection (1) and (2); in respect of matters relatable to any of the entries enumerated in List I and List III in the Seventh Schedule to the Constitution.

Provided where National Disability Rights Authority has reasonable grounds to believe that it is necessary or expedient in the interest of rights of persons with disabilities, it may, by way of resolution supported by not less than two third members of the Governing Board present and voting, inquire into any matter relatable to List II.

106. Power to Conduct Inquiry⁶²

- (1) Where the National Disability Rights Authority receives any petition, complaint or has reasons to believe that there is non-compliance of or violation of any regulations, it may, at any time by order in writing, direct any person specified in the order to investigate and report thereon to the Authority;
- (2) While conducting such investigation, the National Disability Rights Authority shall have all the powers of a civil court trying a suit under the Code of Civil Procedure, 1908, and in particular enjoy the following powers, namely:
 - a. summoning and enforcing the attendance of witnesses and examining them on oath;
 - b. discovery and production of any document;
 - c. receiving evidence on affidavits;
 - d. requisitioning any public record or copy thereof from any court or office;

⁶² As suggested at the Legal Consultation.

- e. issuing commissions for the examination of witnesses or documents;
- f. any other matter which may be prescribed.
- (3) The National Disability Rights Authority may, while investigating a complaint, make written requests for information, interview people, review documents or visit the premises to determine upon the nature of violation or non compliance of regulations;
- (4) Whoever willfully makes a statement which is false in any material particular knowing it to be false or willfully omits to make a material statement in any application, declaration, statement, information or particulars made, required or furnished by or under or for the purposes of any provisions of this Act before National Disability Rights Authority or any order, regulation or direction made or given thereunder by National Disability Rights Authority, shall be punishable under Section 159 of this Act;
- (5) Whoever fails to produce any book, account or other document or to furnish any statement, information or particulars which, under this Act or any order, regulation or direction, whose duty it is to produce or furnish or to answer any question put in pursuance of the provisions of this Act or of any order, regulation or direction made or given thereunder, shall be punishable under Section 160 of this Act.

107. Power to Issue Directions⁶³

- (1) Where during or upon the completion of an enquiry made or caused to be made, the National Disability Rights Authority is satisfied that it is necessary:
 - a. in the interest of promoting rights of persons with disabilities or
 - b. to prevent such acts of omission or commission that are in any manner detrimental to the interests of persons with disabilities; or
 - c. to secure proper compliance with the regulations issued, it may issue such directions or compliance notice, - to an appropriate government or establishment, as may be appropriate in the interests of persons with disabilities.

⁶³ As suggested at the Legal Consultation and added in the National Disability Rights Tribunal

- (2) Any person aggrieved by such directions or compliance notice under sub-section (1), may prefer an appeal before the National Disability Rights Tribunal within a period of 30 days;
- (3) When no such appeal has been preferred and dismissed, such person shall, within a period of 30 days of the expiry of the period of appeal or the date of dismissal, as the case may be, develop a written proposal or enter into a formal agreement on an action plan of compliance;
- **(4)** The National Disability Rights Authority may where it considers feasible, endeavour to ensure compliance of regulations by informal methods of mediation and conciliation:
- (5) The National Disability Rights Authority shall require a regular report on progress with regards to written proposal or formal agreement referred in sub-section (2) and monitor progress thereupon;
- (6) Whoever contravenes or fails to comply with any direction given or compliance notice issued under sub section (2) or formal agreement made under sub section (2) shall be punishable under Sec 161 (1);
- (7) Where any individual, establishment or appropriate government continues the breach of any order, regulation or direction made or given by the National Disability Rights Authority, shall be punishable under Section 161 (2) of this Act.

108. Power to Review and Advice

- (1) Where the National Disability Rights Authority finds that any law, policy or practice directly or indirectly discriminates against or violates the rights of persons with disabilities, it shall advise the appropriate government or establishment referred to under Section 2 (13) (a) to take measures to address the alleged discrimination or violation;
- (2) The appropriate government or aforementioned establishment may consult the National Disability Rights Authority on the advice tendered;
- (3) The appropriate government or aforementioned establishment shall report to the Parliament or State Legislative Assembly, as the case may be, measures taken in pursuance of the advice tendered under sub-section (1).

109. Power to Review, Advice and Report

- (1) Where the National Disability Rights Authority finds that any practice directly or indirectly discriminates against or violates the rights of persons with disabilities, it shall advise the establishment mentioned under Section 2 (13) (b), (c), (d), (e) and (f) to take time bound measures to address the alleged discrimination or violation:
- (2) If the aforementioned establishment fails to take measures under sub-section (1), then the National Disability Rights Authority shall give an opportunity to the establishment to be heard;
- (3) Subsequent to the hearing, the National Disability Rights Authority may forward a report on the establishment's failure to address the alleged discrimination or violation to the National Disability Rights Tribunal for further action.

110. Power to Associate, Co-ordinate with Individuals and Organizations

- (1) The National Disability Rights Authority shall, by way of signing memorandums of understanding, agreements, engaging in active partnerships or through any other procedure associate any person, organization, regulatory authority or authorities whose assistance and advice is desirable for carrying out the purposes of this Act;
- **(2)** The National Disability Rights Authority may constitute Working Groups in specific areas and regions of the Country for the purpose of carrying out its functions and duties.

111. Power to Grant Exemptions

The National Disability Rights Authority may exempt such establishments either generally or for such periods as may be specified, in relation to such matters including any regulation or schemes or programmes formulated under any provision of this Act and subject to such conditions, limitations or restrictions, as may be specified.

112. Establishment of Committees

- (1) The National Disability Rights Authority may establish such Committees as required, for the effective discharge of its functions under this Act and to exercise such powers and functions that may be conferred and such functions as may be assigned to it under this Act;
- (2) The composition, membership, qualifications, terms and conditions of appointment, removal, tenure, removal, tenure, meetings, quorum and resignation of the Committees shall be determined by the regulations.

113. Power of National Disability Rights Authority vis-à-vis the State Disability Rights Authority

- (1) For the purpose of enabling the National Disability Rights Authority to discharge its functions under this Act, it may at any time collect or furnish such information, or direct State Disability Rights Authority to submit to it such statements relating to such information and in such form and within such time as may be specified by the National Disability Rights Authority from time to time;
- (2) The National Disability Rights Authority may, if it considers necessary or expedient in the interest of rights of persons with disabilities, give directions to State Disability Rights Authority either generally or particularly, in respect of any matters relating to or connected with the efficient discharge of functions under this Act;
- (3) The National Disability Rights Authority may, require State Disability Rights Authority to undertake such research, collection data and material for such purposes and within such time as may be specified by the National Disability Rights Authority;
- (4) National Disability Rights Authority may in such as matters of non-compliance or violation of regulations, require State Disability Rights Authority to conduct an inquiry and investigation on its behalf under Section 106 and report thereon;
- (5) The National Disability Rights Authority may associate itself in such manner and for such purposes as required with State Disability Rights Authority;
- (6) The Chairperson of National Disability Rights Authority may convene such number of meetings every year as required, for the purpose of co-ordination,

exchange of information with the Chairpersons of State Disability Rights Authority.

Provided that one such meeting shall be necessarily held every year.

114. Annual Report of the National Disability Rights Authority

- (1) The National Disability Rights Authority shall submit an annual report to the Central Government and may at any time submit special reports on any matter which, in its opinion, is of such urgency or importance that it should not be deferred till submission of the annual report;
- (2) The Central Government shall cause the annual and special reports to be laid before each House of Parliament along with a memorandum of action taken or proposed to be taken on the recommendations of the National Disability Rights Authority and the reasons for non-acceptance of the recommendations, if any.

115. Establishment and Incorporation of State Disability Rights Authority⁶⁴

- (1) There shall be established, for the purposes of promoting, regulating and ensuring the full and equal enjoyment of all human rights and fundamental freedoms of all persons with disabilities, an Authority in each State which shall be called the State Disability Rights Authority;
- (2) State Disability Rights Authority shall be a body corporate having perpetual succession and a common seal, with power, subject to the provisions of this Act, to acquire, hold and dispose of property, both movable and immovable, and to contract, and shall, by the said name, sue or be sued;
- (3) State Disability Rights Authority shall have its head and seat at such place as the State Government may specify;
- (4) State Disability Rights Authority may inquire into violations and non-compliance of regulations formulated by National Disability rights Authority in respect of matters relatable to any of the entries enumerated in List II and List III in the Seventh Schedule to the Constitution.

 $^{^{\}rm 64}$ As suggested by almost all State Consultations.

Provided that if any such matter is already being inquired into by the National Disability Rights Authority or any other Commission duly constituted under any law for the time being in force, the State Disability Rights Authority shall not inquire into the said matter.⁶⁵

116. Composition and Management of State Disability Rights Authority

- (1) State Disability Rights Authority shall consist of a Chairperson and a Board comprising of:
 - a. 5 members of whom at least 3 shall be full-time members who shall be persons with disabilities and shall be appointed by way of an election, by an electoral college that comprises of DPOs, the organizations of parents and families of persons with disabilities and organizations working for the rights of persons with disabilities which are registered under this Act or the National Trust Act, 1999, by such procedure as may be prescribed by the Central Government;

Provided that the members of the first Board shall be nominated by the Governor on obtaining recommendations of the Selection Committee under Section117 from amongst the panel of names forwarded by DPOs, the organizations of parents and families of persons with disabilities and organizations working for the rights of persons with disabilities.

- b. 1 member not below the rank of Joint-Secretary to the State Government and shall be nominated by the State Government, representing the Department of Social Justice and Empowerment (Disability Division);
- c. 1 member nominated by the Confederation Of Indian Industries state office;
- d. 1 member heading the State National Trust office and
- e. a Member-Secretary with requisite knowledge and experience in the field of disability to be appointed in accordance with the procedure prescribed.

⁶⁵ As suggested at the Legal Consultation.

When deemed necessary, for due deliberation on any matter, before the State Disability Rights Authority, the Chairperson may invite a representative of any other Department of the State Government;

- (2) The Board may appoint one of the members to be Vice-Chairperson of the Authority who shall exercise and discharge such powers and functions of the Chairperson as may be prescribed and as may be delegated by the Chairperson;
- (3) The general superintendence, direction and management of the affairs of the State Disability Rights Authority shall vest in the Board, which may exercise all powers and do all acts and things which may be exercised or done by State Disability Rights Authority;
- (4) The Chairperson shall be full time member;
- (5) The Chairperson or any other Member of the State Disability Rights Authority, who is a full-time member, shall not hold any other office of profit;
- (6) The Board shall hold meetings at the head office or any other place at such time as the Chairperson may direct, and shall observe such rules of procedure in regard to the transaction of business at its meetings (including the quorum at its meetings) as may be specified;
- (7) When the Chairperson is unable to discharge the functions of the office owing to absence, illness or any other cause, the Vice-Chairperson shall discharge the functions of the Chairperson until such date that the Chairperson resumes charge;
- (8) All questions which come up before any meeting shall be decided by a majority of votes of the Members present and voting, and in the event of an equality of votes, the Chairperson or the person presiding shall have the right to exercise a second or casting vote;
- (9) No act or proceedings of the Authority shall be questioned or shall be invalidated merely on the ground of existence of any vacancy in, or any defect in, the constitution of the State Disability Rights Authority.

117. Selection Committee for the Chairperson of the State Disability Rights Authority

The Chairperson of State Disability Rights Authority shall be appointed by warrant under the hand and seal of the Governor of the State; provided that every

appointment shall be made after obtaining the recommendations of a Selection Committee from amongst the panel of names forwarded by DPOs, organisations of parents and families of persons with disabilities and organisations working for the rights of persons with disabilities in the State.

- (1) The Selection Committee shall comprise of:
 - a. the Chief Minister of the State;
 - the Leader of Opposition of the State Legislative Assembly;
 - c. the State Minister for Social Justice and Empowerment;
 - d. two experts who shall be persons of integrity and standing who have special knowledge of and experience in the field of disability, law, health, education, management, industry, economics, business, commerce and public affairs.

Provided further that where there is a Legislative Council in a State, the Chairperson of that Council and the Leader of the Opposition in that Council shall also be members of the Committee.

No appointment of Chairperson or other Member shall be invalid merely by reason of any vacancy in the Selection Committee.

118. Term of Office of Chairperson and Members of State Disability Rights Authority

- (1) A person appointed as Chairperson shall hold office for a term of five years from the date of entering office or until the age seventy, whichever is earlier;
- (2) A person appointed as a Member shall hold office for a term of five years from the date on entering upon office and shall be eligible for re-appointment for another term of five years; no person shall be appointed as member of State Disability Rights Authority for more than two consecutive terms or hold office after attaining the age of seventy years.

119. Terms and Conditions of Service of Chairperson and Members of State Disability Rights Authority

The salaries and allowances payable to, and other terms and conditions of service of, the Chairperson and Members shall be such as may be prescribed by the State Government.

Provided that neither the salary and allowances nor the other terms and conditions of service of the Chairperson or a Member shall be varied to his or her disadvantage after his or her appointment

120. Officers and Other Staff of the State Disability Rights Authority

- (1) Subject to such rules as may be made by the State Government in this behalf, the State Disability Rights Authority may appoint such other administrative, technical and scientific staff as it may consider necessary.
- (2) The salaries, allowances and conditions of service of the officers and other staff appointed under sub-section (1) shall be such as may be prescribed by the State Government.

121. Resignation, Removal and Suspension of the Chairperson and Other Members of State Disability Rights Authority

- (1) The Chairperson may resign office by tendering notice in writing to the Governor of the State:
- (2) Any other member may resign office by tendering notice in writing to the Chairperson;
- (3) The Governor of the State may remove the Chairperson from office is he or she:
 - a. has been adjudged a insolvent; or
 - b. has been convicted of an offence which involves moral turpitude; or
 - c. has so abused the position as to render continuance in office prejudicial to public interest.
- **(4)** The Board may, by a resolution passed by two-third majority, remove from office any member, who:
 - a. has been adjudged a insolvent; or
 - b. has been convicted of an offence which involves moral turpitude; or
 - c. has so abused the position as to render his continuance in office prejudicial to the public interest.

- (5) No such member shall be removed from office under clause (c) or clause (d) of sub-section (3) and under clause (c) or clause (d) of sub-section (4) unless given a reasonable opportunity of being heard in the matter;
- (6) In the event of a vacancy in the office of the Chairperson by reason of death, resignation, suspension or removal; the Vice-Chairperson shall act as the Chairperson, until the date on which a new Chairperson, appointed in accordance with the provisions of this Act to fill such vacancy, enters upon office.

122. Application of Certain Provisions Relating to National Disability Rights Authority to State Disability Rights Authority

- (1) The provisions of Sections 105 to 110 and 112 shall apply to a State Disability Rights Authority and shall have effect, subject to the following modifications, namely:
 - a. references to "National Disability Rights Authority" shall be construed as references to "State Disability Rights Authority", except in clause (g) and (l) in sub-section (2) of section 105;
 - b. in sub-section(2) of section 105; clause (n) shall be omitted;
 - c. in sub-section (2) of section 105, in clause (b), the words "Union" shall be omitted:
 - d. references to National Disability Rights Tribunal under sub section (2) of Section 107 and sub section (3) of Section 109 shall be construed as references to State Disability Rights Tribunal.

123. Duties of State Disability Rights Authority vis-a-vis the National Disability Rights Authority

- (1) The State Disability Rights Authority shall furnish the statements, information or particulars called for, to comply with any direction given to it, undertake research and conduct inquiry and report to the National Disability Rights Authority under provisions of this Act;
- (2) The State Disability Rights Authority shall bring to the notice of National Disability Rights Authority, such matters where it is necessary and expedient to do so in the interest of rights of persons with disabilities;

(3) The State Disability Rights Authority shall function in association and extend utmost co-operation to National Disability Rights Authority in furtherance of effective and efficient discharge of its own functions under this Act.

124. Annual and Special Reports of State Disability Rights Authority

- (1) The State Disability Rights Authority shall submit an annual report to the State Government and may at any time submit special reports on any matter which is of such urgency or importance;
- (2) The State Government shall cause the annual and special reports of the State Disability Rights Authority to be laid before each House of State Legislature where it consists of two Houses, or where such Legislature consists of one House, before that House along with a memorandum of action taken or proposed to be taken on the recommendations of the State Disability Rights Authority and the reasons for non-acceptance of the recommendations, if any.

125. Registration of Organizations for Persons with Disabilities

- (1) The State Disability Rights Authority shall register, in accordance with the rules made by the Central Government, association of persons with disabilities or a Disabled Persons' Organizations, association of parents and families of persons with disabilities, or an organization working for the rights of persons with disabilities under this Act or any such person who, being a person with disability, has experience in disability sector or has been a disability rights activist and has completed twenty five years of age;
- (2) Every application for a certificate of registration shall be made to the State Disability Rights Authority in such form and in such manner as may be prescribed;
- (3) On receipt of an application under sub-section (1), the State Disability Rights Authority shall make such enquiries and where it is satisfied that the applicant has complied with the requirements of this Act and the rules made there under, it shall grant a certificate of registration to the applicant. Where the State Disability

- Rights Authority is not satisfied that the requirements have been complied with, it shall give a reasonable opportunity of being heard to the applicant. An order of refusal to grant a certificate shall be accompanied with reasons to be recorded in writing for such non-registration;
- (4) No certificate of registration shall be granted under sub-section (2) unless the organization with respect to which an application has been made is in a position to provide such facilities and maintain such standards as may be prescribed;
- (5) A certificate of registration granted under this section, unless revoked, shall remain in force for such period as may be provided in the rules and may be renewed from time to time for a like period; and shall be in such form and shall be subject to such conditions as may be prescribed;
- **(6)** An application for renewal of a certificate of registration shall be made not less than sixty days before the period of validity;
- (7) The State Disability Rights Authority may, if it has reasonable cause to believe that the holder of the certificate of registration granted under sub-section (3) of has made a statement in relation to any application for the issue or renewal of the certificate which is incorrect or false in material particulars; or committed or has caused to be committed any breach of rules or any conditions subject to which the certificate was granted, it may, after making such inquiry, by order, revoke the certificate:

Provided that no such order shall be made until an opportunity is given to the holder of the certificate to show cause as to why the certificate should not be revoked.

- (8) Where a certificate in respect of an organization has been revoked under subsection (1), such organization shall cease to be in Electoral college from the date of such revocation:
- (9) Every organization which holds a certificate of registration which is revoked under this section shall, immediately after such revocation, surrender such certificate to the State Disability Rights Authority;
- (10) Any person aggrieved by the order of the State Disability Rights Authority refusing to grant a certificate or revoking a certificate may, within such period as may be provided, prefer an appeal to the State Disability Rights Tribunal against such refusal or revocation;
- (11) The order of the State Disability Rights Tribunal on such appeal shall be final.

126. Establishment of National Fund for Persons with Disabilities

- (1) There shall be constituted for the purposes of this Act, a Fund to be called the National Fund for Persons with Disabilities and shall be credited thereto with:
 - a. all moneys payable by banks, corporations, etc., in pursuance of judgment dated 16.04.2004 of the Supreme Court in Civil Appeal No.4655 and 5218 of 2000;
 - b. a sum equivalent to the moneys collected under clause (a) to be contributed by the Central Government;
 - c. all moneys received by the Fund by way of grant, gifts, donations, beneficiations, bequests or transfers; and
 - d. all moneys received by the fund in any other manner or from any other source.
- (2) The money received by the Fund under clauses (a) and (b) of sub section (1) shall constitute the corpus of the Fund which shall be held in trust by the National Disability Rights Authority.
- 127. Dissolution of Erstwhile Fund for Persons with Disabilities and Transfer of their Rights, Assets and Liabilities to the National Fund for Persons with Disabilities.
- (1) On and from the date of the constitution of the National Fund for Persons with Disabilities in accordance with Section 126 the Erstwhile Fund for Persons with Disabilities shall stand dissolved, and on such dissolution:
 - a. all properties and assets, movable and immovable, of, or belonging to the dissolved fund shall vest in the National Fund for Persons with Disabilities;
 - b. all rights and liabilities of the dissolved fund shall be transferred to, and be the rights and liabilities of the National Fund for Persons with Disabilities;
 - c. without prejudice to the provisions of clause (b) all liabilities incurred, all contracts entered into, and all matters and things engaged to be done by, with or for the dissolved Fund, shall be deemed, as the case may be, to

- have been incurred, entered into, or engaged to be done by, with or for, the National Fund for Persons with Disabilities;
- d. all sums of money due to the dissolved Fund immediately before their dissolution shall be deemed to be due to National Fund for Persons with Disabilities:
- e. all suits and other legal proceedings instituted or which could have been instituted by or against the dissolved Fund immediately before their dissolution may be continued or may be instituted by or against the National Fund for Persons with Disabilities.

128. Grants by Central Government

- (1) The Central Government may, after due appropriation made by Parliament by law in this behalf, make to the National Disability Rights Authority grants of such sums of money as the Government may deems fit for being utilised for the purposes of this Act;
- (2) The contribution by Public Sector Undertakings and Public Limited Companies towards the National Disability Fund shall be deemed fulfillment of their Corporate Social Responsibility.

129. Grants by State Government

The State Government may, after due appropriation made by the Legislature by law in this behalf, make to the State Disability Rights Authority grants of such sums of money as the State Government may deems fit for being utilised for the purposes of this Act.

130. Funding Pattern

- (1) The Central Government and State Government shall have concurrent responsibility for providing funds for establishing State Disability Rights Authority;
- (2) The Central Government shall prepare the estimates of capital and recurring expenditure for the establishment of the State Disability Rights Authority;

- (3) The Central Government shall provide to the State Government as grants in aids of revenues, such percentage of expenditure referred to in sub section (2) as it may determine from time to time in consultation with the State government;
- **(4)** Notwithstanding anything contained in sub section (3), it shall be duty of the State Government to provide funds for the establishment of the State Disability Rights Authority.

131. Exemption from tax on wealth and income

Notwithstanding anything contained in the Wealth-tax Act, 1957 (27 of 1957), the Income-tax Act, 1961 (43 of 1961), or any other enactment for the time being in force relating to tax on wealth, income, profits or gains, the National Disability Rights Authority and State Disability Rights Authority shall not be liable to pay wealth-tax, income-tax, or any other tax in respect of their wealth, income, profits or gains derived.

132. Accounts and Audit

- (1) The National Disability Rights Authority shall maintain proper accounts and other relevant records and prepare an annual statement of accounts including the income and expenditure accounts in such form as required by the Comptroller and Auditor-General of India;
- (2) The accounts of the National Fund and National Disability Rights Authority shall be audited by the Comptroller and Auditor General of India at such intervals as may be specified and any expenditure incurred in connection with such audit shall be payable by the National Disability Rights Authority to the Comptroller and Auditor-General of India;
- (3) The Comptroller and Auditor-General of India and any other person appointed in connection with the audit of the accounts of the National Fund and National Disability Rights Authority shall have the same rights, privileges and authority in connection with such audit as the Comptroller and Auditor-General of India has in connection with the audit of Government accounts, and in particular, shall have the right to demand production of books of accounts, connected vouchers, other

- documents and papers and to inspect any of the offices of the National Disability Rights Authority;
- (4) The accounts of the National Fund and National Disability Rights Authority as certified by the Comptroller and Auditor-General of India or any other person appointed in this behalf, together with the audit report thereon, shall be forwarded annually to the Central Government which shall cause the same to be laid before each House of Parliament.

133. Accounts and Audit of State Disability Rights Authority

- (1) The State Disability Rights Authority shall maintain proper accounts and other relevant records and prepare an annual statement of accounts in such form as may be prescribed by the State Government in consultation with the Comptroller and Auditor-General of India;
- (2) The accounts of the State Disability Rights Authority shall be audited by the Comptroller and Auditor-General at such intervals as may be specified and any expenditure incurred in connection with such audit shall be payable by the State to the Comptroller and Auditor-General;
- (3) The Comptroller and Auditor-General or any person appointed in connection with the audit of the accounts of the State Disability Rights Authority under this Act shall have the same rights and privileges and the authority in connection with such audit as the Comptroller and Auditor-General generally has in connection with the audit of Government accounts and, in particular, shall have the right to demand the production of books, accounts, connected vouchers and other documents and papers and to inspect any of the offices of the State Disability Rights Authority;
- (4) The accounts of the State Disability Rights Authority, as certified by the Comptroller and Auditor-General or any other person appointed in this behalf, together with the audit report thereon, shall be forwarded annually to the State Government by the State Disability Rights Authority and the State Government shall cause the audit report to be laid, as soon as may be after it is received, before the State Legislature.

PART IV: GRIEVANCE REDRESSAL

134. District Disability Rights Tribunal:

- (1) There shall be established District Disability Rights Tribunals at the district level in each State;
- (2) The Tribunal shall consist of two members:
 - a. a presiding officer being an advocate who has practiced for not less than seven years;
 - b. a member who is an experiential expert or a subject expert with experience in disability human rights or human rights;
- (3) The above appointments shall be in accordance in with prescribed procedure;
- **(4)** The salary allowances and other emoluments of the Presiding officer shall be of senior judge, civil division and that of the member Junior Judge civil division.

135. Jurisdiction of the District Disability Rights Tribunal

The tribunal shall have jurisdiction over such number of Revenue Districts as may be notified keeping in view the territorial expanse of districts in each state and the demographic concentration of persons with disabilities;

136. Panel of Disability Rights Defenders

- (1) Each District Disability Rights Tribunal shall have a panel of three advocates attached to it as Disability Rights Defenders;
- (2) The Disability Rights Defenders shall be attached to the respective Tribunal on a full-time basis and will assist the Tribunal in the conduct of the proceedings before it;
- (3) Advocates with five years practicing experience at the bar shall be eligible for appointment as Disability Rights defenders;
- (4) The pay and allowances of Disability Rights Defenders shall be as prescribed.

137. Proceedings before the District Disability Rights Tribunal

- (1) Any person with disability aggrieved by the deprivation of the rights guaranteed under this Act and the National Trust Act can petition the District Disability Rights Tribunal in person or by post by such procedure as may be prescribed;
- (2) The tribunal shall tour the entire area under its jurisdiction and hear dispose of matters at mandal, tehsil or taluk level headquarters closest to the place where the petitioner with disability resides;
- (3) On receipt of a complaint, the District Disability Rights Tribunal shall provide a copy of the complaint to the opposite party mentioned in the complaint directing such party to respond within a period of thirty days;
- (4) The District Disability Rights Tribunal shall, after considering the written response to the complaint, communicate to both parties the place where the proceeding is to be conducted and give a reasonable opportunity to both the parties to present their case in person or through legal counsel.

Provided that persons with disabilities shall be assisted by the Disability Rights Defenders attached to the Tribunal to present their case.

(5) The decision of the District Disability Rights Tribunal shall be final.

138. State Disability Rights Tribunal

- (1) There shall be established State Disability Rights Tribunal in each state;
- **(2)** The tribunal shall have three members:
 - a. A Presiding Officer from amongst the District Judges in the State with the concurrence of the Chief Justice of the High Court for a period of three years or a person who has served as a presiding officer in one of the District Disability Tribunals in the state for at least a period of 5 years;
 - Persons who have either held or were eligible to hold the office of the Commissioner for Persons with Disabilities under the Persons with Disabilities (Equal Opportunities, Protection of Rights & Full Participation) Act, 1995; member
 - c. A graduate in humanities, social sciences, psychology, special education or social work with at least 5 years experience in disability human rights or human rights; or a person who has served as a member of one of the District Disability Rights Tribunals in the state for at least a period of 5 years; member

(3) The appointments shall be made in accordance with such procedure as may be prescribed.

139. Jurisdiction of the State Disability Rights Tribunal

- (1) If the District Disability Rights Tribunal is of the opinion that any matter before it involves important question of law then it may refer the same to the State Disability Rights Tribunal for its decision;
- (2) If there is a difference of opinion between the Presiding Judge and Member of the District Disability Rights Tribunal then the same shall be referred to the State Disability Rights Tribunal for its decision;
- (3) If the State Disability Tribunal is of the view that any case has been pending before the District Disability Rights Tribunal for an unduly long time then it may withdraw the same to its file and dispose it off;
- (4) If the State Disability Rights Tribunal is of the opinion that a similar question is pending for decision before more than one District Disability Rights Tribunal under its jurisdiction then it may withdraw all those cases from different Tribunals to its file and dispose them off in accordance with law;

Provided that the State Disability Rights Tribunal may either exercise this power suo motu or on application of any interested person

- **(5)** The State Disability Rights Tribunal may either suo motu or on application take cognizance, of any large scale or systemic infringement of rights of persons with disabilities and pass appropriate orders or directions;
- (6) Any person aggrieved by a compliance notice or direction of State Disability Rights Authority made under this Act, and the attendant regulations may appeal against the same to the State Disability Rights Tribunal within a period of 30 days.

140. Proceedings before the State Disability Rights Tribunal

(1) The State Disability Rights Tribunal shall devise its own procedure to conduct its proceedings provided that such procedure is in accordance with the principles of natural justice;

- (2) The State Disability Rights Tribunal shall ordinarily sit as a full panel of three members:
 - However if the presiding officer is on leave he may authorize the other two members to function as a Tribunal. In the event of a member being absent by reason of leave or post being vacant the presiding officer and the other member will function as the Tribunal
- (3) In the event of difference of opinion, the majority opinion shall prevail;
- (4) The decision of the State Disability Rights Tribunal shall be final.

141. National Disability Rights Tribunal

- (1) There shall be established a National Disability Rights Tribunal in New Delhi;
- (2) The tribunal shall have three members:
 - a. A Presiding Officer from amongst the High Court Judges with the concurrence of the Chief Justice of India for a period of three years or a person who has served as a presiding officer in one of the State Disability Rights Tribunals for at least a period of 5 years;
 - b. Persons who have either held or were eligible to hold the office of the Chief Commissioner or for Persons with Disabilities under the Persons with Disabilities (Equal Opportunities, Protection of Rights & Full Participation) Act, 1995: member
 - c. A graduate in humanities, social sciences, psychology, special education or social work with at least 7 years experience in disability human rights or human rights as member or a person who has served as a member of one of the State Disability Tribunals for at least a period of 5 years: member
- **(3)** The appointments shall be made according to prescribed procedure.

142. Jurisdiction of the National Disability Rights Tribunal

(1) If the State Disability Rights Tribunal is of the opinion that any matter before it involves an important question of law which merits the consideration of the National Disability Rights Tribunal then it may refer the same to the National Disability Rights Tribunal for its decision; and upon such reference the National Disability Rights Tribunal shall entertain the matter and dispose it of; (2) If the National Disability Rights Tribunal is of the opinion that a similar question is pending for decision before more than one State Disability Rights Tribunal under its jurisdiction then it may withdraw all those cases from different Tribunals to its file and dispose them of in accordance with law;

Provided that the National Disability Rights Tribunal may either exercises this power suo motu or on application of any interested person

- (3) The National Disability Rights Tribunal shall either suo motu or on application, take cognizance, of any large scale or systemic infringement of rights of persons with disabilities and pass appropriate orders or directions;
- (4) Any person aggrieved by a compliance notice or direction of National Disability Rights Authority made under this Act, and the attendant regulations may appeal against the same to the National Disability Rights Tribunal within a period of 30 days.

143. Powers and Functions of the Disability Rights Tribunals

- (1) The Disability Rights Tribunals shall decide both questions of law and facts that may be raised before it and if it is satisfied that there has been a violation of any right guaranteed under this Act or violation of any provision of this Act, it shall issue an order to the opposite party directing one or more of the following things:
 - a. to remedy the infringement by removing the discriminatory practice or scheme;
 - b. to discontinue any activity that is violative of the provisions of this Act;
 - c. to award compensation to the victim for any loss or injury suffered;
 - d. to provide adequate costs to the parties.
- (2) Every order made by the Disability Rights Tribunals shall be signed by the members conducting the proceedings and published regularly on its website.

144. Powers of the Disability Rights Tribunals

(1) For the purposes of this section, the Disability Rights Tribunals shall have the same powers as that of a Civil Court under the Code of Civil Procedure 1908 while trying a suit in respect of the following matters, namely:

- a. summoning and enforcing the attendance of any person and examining him on oath;
- b. requiring the discovery and production of documents;
- c. receiving evidence on affidavits;
- d. subject to the provisions of sections 123 and 124 of the Indian Evidence Act, 1872, (1 of 1872) requisitioning any public record or document or copy of such record or document from any office;
- e. issuing commissions for the examination of witnesses or documents;
- f. reviewing its decisions;
- g. dismissing an application for default or deciding it ex parte;
- h. setting aside any order of dismissal of any application for default or any order passed by it ex parte; and
- i. any other matter which may be prescribed by the Concerned Government.

145. Execution of the Orders of the Disability Rights Tribunals

- (1) An order or decision of the Tribunal under this Act shall be executable by the Tribunal as a decree of a civil court, and for this purpose, the Tribunal shall have all the powers of a civil court;
- (2) Every order passed by the District Disability Rights Tribunals shall be implemented within a period of four months and the authority to implement the same shall be the Revenue Divisional Officer or Sub Divisional Magistrate as the case may be having jurisdiction over the area where the person in whose favour the order is made resides. The orders of the State and National Disability Rights Tribunals shall be implemented by District Collector, District Magistrate or Deputy Commissioners as the case may be having jurisdiction over the area where the person in whose favour the order is made resides.

146. Conditions of Service of Members

The salaries and allowances payable to, and other terms and conditions of service of the members of the Disability Rights Tribunals, including the Disability Rights Defenders, shall be such as may be prescribed.

147. Officers and other Staff of Disability Rights Tribunals

- (1) The central government in the case of the National Disability Rights Tribunal and the respective state governments in the case of other tribunals shall be responsible for providing the staff and necessary infrastructure for the effective functioning of the Tribunals;
- (2) Subject to such Rules as may be made by the Central Government in this behalf, the Disability Rights Tribunals may appoint such other technical and scientific staff as it may consider necessary;
- (3) The salaries, allowances and conditions of service of the officers and other staff shall be such as may be prescribed.

PART V: OFFENCES AND PENALITIES

148. Cognizance of Offences and Jurisdiction of Courts

- (1) The Court of Metropolitan Magistrate or Judicial Magistrate first class having jurisdiction over the area in which the offence has taken place alone shall have the power to take cognizance of the offences under this Act;
- (2) No Court shall take cognizance of an offence under this Act except:
 - a. on a written complaint of the aggrieved person or a friend or a relative or a registered organization as so defined in section 2(31) of this Act; or
 - b. on a written complaint for any offence committed under sub section (4) or (5) of Section 106 or under sub-section (5) or (6)of Section 107 by the National Disability Rights Authority or the State Disability Rights Authority, as the case may be or any person authorized by or on behalf of National Disability Rights Authority or State Disability Rights Authority as the case may be;
- (3) In addition to any sentence awarded or fine imposed under this Act, the Court may provide for the payment of appropriate monetary compensation proportionate to the injury caused or harm and injury suffered by the aggrieved person.

149. Offences by Establishments

(1) Where any offence under this Act has been committed by an establishment, every person who, at the time the offence was committed, was the appointed

head or was directly in charge of, and was responsible to the establishment for the conduct of its business, as well as the establishment, shall be deemed to be guilty of the offence, and shall be liable to be proceeded against and punished accordingly;

Provided that nothing contained in this sub-section shall render any such person liable to any punishment provided in this Act, if he proved that the offence was committed without his knowledge or that all due diligence was exercised to prevent the commission of such offence.

(2) Notwithstanding anything contained in sub-section (1), where an offence under this Act has been committed by an establishment and it is proved that the offence has been committed with the consent or connivance of, or is attributable to any neglect on the part of any other officer of the establishment, such person shall also deemed to be guilty of that offence and shall be liable to be proceeded against and punished accordingly.

150. General Penalty for Offences

- (1) Whoever fails to comply or contravenes any of the provisions of this Act, or the rules made or orders or directions issued hereunder, shall, in respect of each such failure or contravention, be punishable with imprisonment for a term which may extend to six months or with fine which may extend to fifty thousand rupees, or with both. In case the failure or contravention continues, with additional fine which may extend to one thousand rupees for every day to which such failure or contravention continues after the conviction for the first such failure or contravention:
- (2) If the failure or contravention referred to in sub-section (1) continues beyond a period of one year after the date of conviction, the offender shall be punishable with imprisonment for a term which may extend to four years.

151. Punishment for Injury to Person with Disability

Whoever voluntarily injures, damages or interferes with the use of any limb or sense or faculty of a person with disability, permanently or temporarily shall be punishable with imprisonment for a term which shall not be less than six months

but which may extend to eight years, and with fine, or with both.

152. Penalty for Wrongful Medical Procedures

- (1) Whoever performs, conducts or directs any medical procedure to be performed on a person with disability which leads to or is likely to lead to infertility in contravention of Section 30 shall be punishable with imprisonment for a period not exceeding seven years and with fine;
- (2) Any person acting as a care-giver of the person with disability, whether as parent or guardian or in any other capacity, lawful or unlawful, who does any act to facilitate, or negligently fails to prevent such medical procedure from being performed, shall be punishable with imprisonment which may extend to five years and with fine.

153. Penalty for Forceful Termination of Pregnancy

- (1) Whoever performs conducts or directs any medical procedure to be performed on a woman with disability which leads to or is likely to lead to termination of pregnancy without her express consent shall be punishable with imprisonment for a period not exceeding ten years and with fine;
- (2) Any person acting as a care-giver of the woman with disability, whether as parent or guardian or in any other capacity, lawful or unlawful, who does any act to facilitate, or negligently fails to prevent such medical procedure from being performed, shall be punishable with imprisonment which may extend to seven years and with fine.

154. Penalty for Hate Speech

Whoever voluntarily or knowingly commits or abets the offence of uttering hate speech shall be inducted into one or more different community service programmes, or shall be punishable with fine, or with both.

Explanation

Community service programme means a programme developed by the appropriate government, which shall include, but not be limited to engaging individuals with working in various institutions housing persons with disabilities and participating in

awareness programmes relating to persons with disabilities.

155. Penalty for Denial of Food and Fluids

Whoever voluntarily or knowingly denies food or fluids to any person with disability or aids or abets in such denial shall be punishable with punishable with imprisonment for a term which shall not be less than six months but which may extend to three years, and with fine.

156. Penalty for Contravention of Accessibility Norms for Built Infrastructure

- (1) Any individual or any organization or establishment which fails to comply with the provisions under Section 83 shall be punished with:
 - a. Suspension of the supply of water and electricity till the accessibility requirements are fulfilled; or
 - b. Cancellation, revocation or suspension of the license of the builder; or
 - c. Fine which may extend to five percent of the market value or the estimated market value, as the case may be, of the building; or
 - d. A combination of two or more of the above.
- (2) If the offender fails to comply with the provisions of Section 83 within a period of six months from the date of conviction, the offender shall be punished with an additional fine which may extend up to one lakh rupees for every day during which such failure continues after the conviction for the first such failure.

157. Penalty for Contravention of Duties by Service Providers

- (1) Any service provider who fails to comply with the provisions under Section 84 shall be penalized with cancellation, revocation or suspension of service license or manufacturing; or with fine, which may extend up to one lakh rupees or with both;
- (2) If the offender fails to comply with the provisions of Section 84 within a period of six months from the date of conviction, the offender shall be penalised with additional fine which may extend up to five thousand rupees for every day during which such failure continues after the conviction for first such failure.

158. Penalty for Fraudulently Availing Benefit Meant for Persons with Disabilities

Whoever fraudulently avails or attempts to avail any benefit meant for persons with disabilities shall be punishable with imprisonment for a term which may extend to two years or with fine which may extend to twenty thousand rupees or both.

159. Penalty for Willfully Making False Statement

Whoever in any application, declaration, statement, information or particulars made, required or furnished by or under or for the purposes of any provisions of this Act before National Disability Rights Authority or State Disability Rights Authority as the case may be, or any order, regulation or direction made or given thereunder by National Disability Rights Authority or State Disability Rights Authority, willfully makes a statement which is false in any material particular knowing it to be false or willfully omits to make a material statement shall be punishable with imprisonment for a term which may extend to three years and shall also be liable to fine which may extend to rupees one lakh.

160. Penalty for Failure to Furnish Information

Whoever fails to produce any book, account or other document or to furnish any statement, information or particulars which, under this Act or any order, regulation or direction made or given thereunder, whose duty it is to produce or furnish or to answer any question put in pursuance of the provisions of this Act or of any order, regulation or direction made or given thereunder, shall be punishable with fine which may extend to fifteen thousand rupees in respect of each offence and if in case of persistence in such failure or refusal, with further fine which may extend to one hundred rupees for every day, after the first during which the offence continues.

161. Penalty for Failure to Comply with Directions

(1) Whoever contravenes or fails to comply with any direction given or compliance notice issued or formal agreement made under Section 107 by the National Disability Rights Authority or State Disability Rights Authority as the case may be,

- shall be punishable with imprisonment for a term which shall not be less than one year but which may extend to five years and with fine which shall not be less than one lakh rupees but which may extend to five lakh rupees;
- (2) Where any individual, establishment or appropriate government continues the breach of any order, regulation or direction made or given by the National Disability Rights Authority or State Disability Rights Authority as the case may be, every such individual or such person who at the time the breach continued was in charge of, and was responsible to, the establishment or appropriate government for the conduct of its business and/or management, functioning, administration or had been the Managing Director, Secretary, shall be deemed to be guilty of the non-compliance and shall be liable to be proceeded against and punished accordingly.

Provided that nothing contained in this sub-section shall render any such person liable to any punishment provided in this Act, if it is proved that the offence was committed without knowledge of such person or that all due diligence had been exercised to prevent the commission of such offence.

PART VI: MISCELLANEOUS

162. Effect of this Act on Other Laws

- (1) The provisions of this Act shall have effect notwithstanding anything inconsistent therewith contained in any other law for the time being in force and to the extent of such inconsistency that other law shall be deemed to have no effect.
- (2) The provisions of this Act or the rules made there under shall be in addition and not in derogation of any other legislation, rules, orders or instructions which provides any entitlement or benefit to persons with disabilities.

163. Action taken in Good Faith

No suit prosecution or other legal proceeding shall lie against any person for anything which is in good faith done or intended to be done under this Act or any rule there made under

164. Provision for Mandatory Review

- (1) There shall be a mandatory review of the functioning of this Act once in every ten years after all parts of this legislation have come into force in the entire country.
- (2) Such review shall evaluate whether the legislative provisions are meeting the rights requirements of persons with disabilities; and whether any additions, modifications or deletions need to be made in the legislation in furtherance of the rights of persons with disabilities.

165. Removal of Difficulties Clause

(1) If any difficulty arises in giving effect to the provisions of this Act, the Central Government may, by order published in the Official Gazette, make such provisions, not inconsistent with the provisions of this Act as appear to it to be necessary or expedient for removing the difficulty;

Provided that no such order shall be made after the expiry of the period of two years from the date of commencement of this Act.

(2) Every order made under this section shall as soon as may be after it is made, be laid before each House of parliament.

166. Power of Central Government to Make Rules

- (1) The Central Government may, by notification make rules, for the effective implementation and the purposes of this Act;
- **(2)** Without prejudice to the generality of sub-section (1), the Central government shall make rules relating to:
 - a. The issuance of driving licenses to persons with disabilities considering their accessibility requirements;
 - b. Measures and conditions relating to the Equal Opportunity Policy to be formulated by establishments;
 - c. Areas and limits for establishment of neighbouring schools;
 - d. The minimum qualification criteria for the persons with disabilities seeking admission to higher educational institutions;

- e. Accessibility to infrastructure, facilities, services, information and communication technologies and personal mobility bearing in mind the diversity of disability and principles of universal design;
- f. The form of records to be maintained pertaining to allocation and utilization of funds for women and girls with disabilities in various schemes and programmes;
- g. The scales for providing compensation and reimbursement of medical expenses;
- h. Qualifications and requisite number of professionals to provide services to persons with disabilities;
- i. Elimination of discrimination in matters of employment, work and occupation;
- j. Qualifications and training of educators;
- k. procedure to constitute the Executive Board of National Disability Rights Authority;
- I. deputation of government officers to the Executive Board;
- m. the salaries and allowances and other terms and conditions of service of the Chairperson and members of the National Disability Rights Authority;
- n. Powers and functions of the Vice-Chairperson of the National Disability Rights Authority;
- o. the qualifications, duties and powers and terms of conditions of service of the Member-Secretary to the National Disability Rights Authority;
- the salaries, allowances and conditions subject to which other administrative, technical and scientific staff may be appointed by the National Disability Rights Authority;
- q. any other power of a civil court required to be prescribed under clause (vi) of Section 99:
- r. the qualifications for any person contesting the election to the membership to the Disability Rights Authority bearing in mind the experience and contribution required for such posts;
- s. rules of procedure for the registration of members of Electoral college;
- t. the form and manner of application for registration of members;
- u. the facilities and standards to be maintained by the organization;
- v. Reservation of posts within the National Disability Rights Authority for ensuring diversity and representation of various forms of disability;

- w. the appointment, salary and allowances payable to and the other terms and conditions of service of the Presiding Officer and members of the National Disability Rights Tribunal;
- x. the salary and allowances and other conditions of service of the officers and employees of the National Disability Rights Tribunal;
- y. the salaries, allowances and conditions subject to which other administrative, technical and scientific staff may be appointed by the National Disability Rights Tribunal;
- z. the additional matters in respect of which the National Disability Rights
 Tribunal may exercise the powers of a Civil Court;
- aa. the form in which and the time at which service of notices to any person or to the Central Government for the purpose;
- bb. the manner of delivery of every notice, order or document to be served;
- cc. any other power of a civil court required to be prescribed;
- dd. Procedure relating to the preparation and publication of the annual reports of the National Fund, National Disability Rights Authority and the State Disability Rights Authority;
- ee. Any other matter incidental or ancillary for the enforcement of this Act.
- (3) Prior to the formation of the rules, the central government shall make available the draft of the rules in accessible formats to the members of the public and invite their suggestions and objections to the same;
- (4) Every rule made by the Central Government under this Act shall be laid, as soon as may be after it is made, before each house of Parliament, while it is in session, for a total period of 30 days which may be comprised in one session or in two or more successive sessions and if, before the expiry of the session immediately following the session or the successive sessions aforesaid, both houses agree in making any modification in the rule or both houses agree that the rule should not be made. The rule shall thereafter have effect only in such modified form, or be of no effect as the case may be; so, however, that any such modification or annulment shall be without prejudice to the validity of anything previously done under that rule.

167. Power of the State Government to Make Rules

- (1) The State Government shall have the power to make rules for the effective implementation and the purposes of this Act;
- (2) Without prejudice to the generality of sub-section (1), the State Government shall make rules relating to:
 - a. The form of records to be maintained pertaining to allocation and utilization of funds for women and girls with disabilities in various schemes and programmes;
 - b. the salaries and allowances and other terms and conditions of service of the Chairperson and members of the State Disability Rights Authority;
 - c. powers and functions of the Vice-Chairperson of the State Disability Rights Authority;
 - d. the salary and allowances and other conditions of service of the officers and employees of the State Disability Rights Authority;
 - e. the salaries, allowances and conditions subject to which other administrative, technical and scientific staff may be appointed by the State Disability Rights Authority;
 - f. the qualifications, duties and powers and terms of condition of service of the Member-Secretary to the State Disability Rights Authority;
 - g. any other power of a civil court required to be prescribed;
 - h. the appointment, salary and allowances payable to and the other terms and conditions of service of the Presiding Officer and members of the State Disability Rights Tribunal;
 - the salaries, allowances and conditions subject to which other administrative, technical and scientific staff may be appointed by the State Disability Rights Tribunal;
 - j. the additional matters in respect of which the State Disability Rights Tribunal may exercise the powers of a civil court;
 - k. the form in which and the time at which service of notices to any person or to the State Government for the purpose;
 - I. the manner of delivery of every notice, order or document to be served;
 - m. Procedure relating to the preparation and publication of the annual reports of the State Disability Rights Authority;
 - n. procedure of establishing Authorities for issuing disability certificates;
 - o. Rules relating to recognition of "National Disability Certificate and
 - p. Any other matter incidental or ancillary for the enforcement of this Act.

- (3) Prior to the formation of the rules, the State Government shall make available the draft of the rules in accessible formats to the members of the public and invite their suggestions and objections to the same.
- (4) Every rule made by the State Government under this section shall be laid, as soon as may be after it is made, before each house of the state legislature where it consists of two houses, or where such legislature consists of one house, before that house.

168. Powers of National Disability Rights Authority to Make Regulations

- (1) The National Disability Rights Authority may, by notification make regulations consistent with this Act and the Rules for the effective implementation of the purposes of this Act, in respect of the matters relatable to any of the entries enumerated in List I and List III in the Seventh Schedule to the Constitution;
- (2) Without prejudice to the generality of sub-section (1), the National Disability Rights Authority shall make regulations relating to:
 - a. Standards for accessibility to physical environment, transportation, information and communication technologies including appropriate technology and systems and other facilities and services with due consideration to the diversity of disability and principles of universal design;
 - b. Standards of facilities for Service animal training.
 - c. Standards for accessibility for existing infrastructure and premises;
 - d. Standards for accessibility for all service providers;
 - e. Training and educational qualifications for teachers having regard to the support requirements of persons with disabilities;
 - f. Procedure relating to matters;
 - g. The establishment, powers and duties of Committees formed by National Disability Rights Authority;
 - h. The composition, membership, qualifications, terms and conditions of appointment, removal, tenure, meetings, quorum and resignation of the Committees formed by the National Disability Rights Authority;
 - i. The rules of procedure for transaction of business;
 - j. The rules of procedure for convening meetings of Chairpersons;
 - k. Conditions, limitations and restrictions of exemptions;

- The form in which and the time at which the meetings to be called;
- m. The form and manner in which and the time at which research needs to be undertaken:
- n. The manner in which and the time at which meetings of National Disability Rights Authority to be convened;
- o. Any other matter incidental or ancillary for the enforcement of this Act.
- (3) Prior to the finalization of the regulations, the National Disability Rights Authority shall make available the draft rules in accessible formats to the members of the public and invite their suggestions and objections to the same.

169. Power of the State Disability Rights Authority to make Regulations

(1) The State Disability Rights Authority shall have the power to issue regulations which are in conformity with the Act and the Rules for the effective implementation of the purposes of this Act, in respect of the matters relatable to any of the entries enumerated in List II and List III in the Seventh Schedule to the Constitution;

Provided that where any regulations has already been issued in any such matter pertaining to List III by the National Disability Rights Authority, such regulations shall prevail over the regulations of the State Disability Rights Authority pertaining to the said matter.

- **(2)** Without prejudice to the generality of sub-section (1), the State Disability Rights Authority shall make regulations relating to:
 - a. Establishment of such Committees as may be required for the efficient performance of its functions;
 - b. The composition, membership, qualifications, terms and conditions of appointment, removal, tenure, meetings, quorum and resignation of the Committees formed by the State Disability Rights Authority;
 - c. Any other matter incidental or ancillary for the enforcement of this Act.
- (3) Prior to the finalization of the regulations, the State Disability Rights Authority shall make available the draft rules in accessible formats to the members of the public and invite their suggestions and objections to the same.

170. Repeal and Saving

- (1) The Persons with Disabilities (Equal Opportunity Protection of Rights and Full Participation) Act of 1995 (Act No.1 of 1996) is hereby repealed;
- (2) Notwithstanding such repeal anything done or any action taken under either of the said acts insofar as such thing or action is not inconsistent with the provisions of this Act be deemed to be have been done or taken under the corresponding provisions of this Act and shall continue in force until superseded by anything done or any action taken under this Act.

SCHEDULE 1

List of Disabilities

- (1) Autism Spectrum Conditions / Autism Spectrum Disorders refers to a neurological condition typically appearing in the first three years of life, that significantly affects a person's ability to communicate, understand relationships and relate to others, and is frequently associated with unusual or stereotypical rituals or behaviours. Autism is often accompanied by learning disability, severe sensory defensiveness, motor coordination difficulties, intellectual impairment, limitations in adaptive behavior, and seizure disorder, among others.
- (2) 'Blindness' refers to a condition where a person has any of the following conditions, namely:
 - a. Total absence of sight; or
 - b. Visual acuity not exceeding 6/60 or 20/200 (Snellen) in the better eye with correcting lenses; or
 - c. Limitation of the field of vision subtending an angle of 20 degree or worse.
- (3) 'Cerebral Palsy' refers to a group of chronic conditions affecting body movements and muscle coordination. It is caused by damage to one or more specific areas of the brain, usually occurring during fetal development or infancy. It also can occur before, during or shortly following birth.
- (4) 'Chronic neurological conditions' refers to a condition that has its origin in some part of the person's nervous system lasting for a long period of time or marked by frequent recurrence.
- **(5)** 'Deafblindness'refers to a condition in which people may have a combination of hearing and visual impairments causing severe communication, developmental, and educational problems. Deafblindness may include:
 - Moderate to profound hearing and significant visual impairments;
 - Moderate to profound hearing and significant visual impairments and other significant disabilities;
 - Central processing problems of vision and hearing;
 - Progressive sensory impairments/ significant visual impairment; and

- Possible loss of auditory processing mechanisms (associated with severe physical disability or severe cognitive disability) and severe communication delay"
- **(6)** 'Dwarfism' is characterized by disproportionate shortness of stature measured by height to age. It is usually it is less than three standard deviation of the average height to age.
- (7) 'Hemophilia' refers to an inheritable disease, usually affecting only males but transmitted by women to their male children, characterized by loss or impairment of the normal clotting ability of blood so that a minor wound may result in fatal bleeding
- **(8)** 'Hearing Impairment' refers to loss of 60 decibels or more in the better ear in the conversational range of frequencies; such impairment in hearing, whether permanent or fluctuating, that hinders the communication with others.
- (9) 'Hard of Hearing' refers to those persons with hearing impairments with a permanent or fluctuating hearing loss which permits the use of the auditory channel for a certain amount of speech/language and information gathering functions.
- (10) 'Intellectual Disability' refers to a disability characterized by significant limitations both in intellectual functioning (reasoning, learning, problem solving) and in adaptive behavior, which covers a range of everyday social and practical skills.
- (11) 'Leprosy cured' person refers to any person who has been cured of leprosy but is suffering from:
 - a. Loss of sensation in hands or feet as well as loss of sensation and paresis in the eye and eye-lid but with no manifest deformity;
 - b. Manifest deformity and paresis but having sufficient mobility in their hands and feet to enable them to engage in normal economic activity;
 - c. Extreme physical deformity as well as advanced age which prevents him/her from undertaking any gainful occupation, and the expression "leprosy cured" shall construed accordingly.
- (12) 'Locomotor Disability' refers to a person's inability to execute distinctive activities associated with movement of self and objects resulting from affliction of musculoskeletal and/or nervous system
- (13) 'Low-vision' refers to a condition where a person has any of the following conditions, namely:

- a. Visual acuity not exceeding 6/18 or 20/60 and less than 6/60 or 20/200 (Snellen) in the better eye with correcting lenses; or
- b. Limitation of the field of vision subtending an angle of more than 20 degree and up to 40 degree.
- (14) 'Mental illness' refers to a disorder of mood, thought, perception, orientation or memory which causes significant distress to a person or impairs a person's behavior, judgment and ability to recognize reality or impairs the person's ability to meet the demands and activities of daily life and includes mental conditions associated with the abuse of alcohol and drugs.
- refers to two or more impairments occurring at the same time the combination of which causes significant needs. The term does not include deafblindness.
- (16) 'Muscular Dystrophy' A group of hereditary genetic muscle disease that weakens the muscles that move the human body. People with MD have incorrect or missing information in their genes, which prevents them from making the proteins they need for healthy muscles. It is characterized by progressive skeletal muscle weakness, defects in muscle proteins, and the death of muscle cells and tissue.
- (17) 'Multiple Sclerosis' An inflammatory, nervous system disease in which the myelin sheaths around the axons of nerve cells of the brain and spinal cord are damaged, leading to demyelination and affecting the ability of nerve cells in the brain and spinal cord to communicate with each other.
- (18) 'Specific Learning Disabilities' refers to a heterogeneous group of conditions wherein there is a deficit in processing language, spoken or written, that may manifest itself as a difficulty to comprehend, speak, read, write, spell, or to do mathematical calculations. The term includes such conditions as perceptual disabilities, dyslexia, dysgraphia, dyscalculia, dyspraxia and developmental aphasia. 'Dyslexia' refers to difficulties in reading, writing and spelling; Dysgraphia' refers to difficulties in the ability to write, primarily in terms of handwriting and content; 'Dyscalculia' refers to difficulties in the ability to learn class/age appropriate mathematics; 'Dyspraxia' refers to difficulties of motor coordination usually apparent in childhood

- (19) 'Speech impairment' refers to an impairment of speech articulation, voice, fluency, or the impairment of language comprehension and/oral expression or the impairment of the use of a spoken or other symbol system.
- (20) 'Thalassemia' refers to a group of inherited disorders characterized by reduced or absent amounts of hemoglobin.